

Review



Intelligent Systems Using Sensors and/or Machine Learning to Mitigate Wildlife–Vehicle Collisions: A Review, Challenges, and New Perspectives

Irene Nandutu D, Marcellin Atemkeng *D and Patrice Okouma D

Department of Mathematics, Rhodes University, Artillery Rd., Grahamstown 6139, South Africa; irenenanduttu@gmail.com (I.N.); p.okouma@ru.ac.za (P.O.)

* Correspondence: m.atemkeng@ru.ac.za

Abstract: Worldwide, the persistent trend of human and animal life losses, as well as damage to properties due to wildlife-vehicle collisions (WVCs) remains a significant source of concerns for a broad range of stakeholders. To mitigate their occurrences and impact, many approaches are being adopted, with varying successes. Because of their increased versatility and increasing efficiency, Artificial Intelligence-based methods have been experiencing a significant level of adoption. The present work extensively reviews the literature on intelligent systems incorporating sensor technologies and/or machine learning methods to mitigate WVCs. Included in our review is an investigation of key factors contributing to human-wildlife conflicts, as well as a discussion of dominant state-of-the-art datasets used in the mitigation of WVCs. Our study combines a systematic review with bibliometric analysis. We find that most animal detection systems (excluding autonomous vehicles) are relying neither on state-of-the-art datasets nor on recent breakthrough machine learning approaches. We, therefore, argue that the use of the latest datasets and machine learning techniques will minimize false detection and improve model performance. In addition, the present work covers a comprehensive list of associated challenges ranging from failure to detect hotspot areas to limitations in training datasets. Future research directions identified include the design and development of algorithms for real-time animal detection systems. The latter provides a rationale for the applicability of our proposed solutions, for which we designed a continuous product development lifecycle to determine their feasibility.

Keywords: wildlife–vehicle collisions; intelligent systems; sensor; machine learning; human–wildlife; human behavior; animal behavior; machine learning datasets; animal detection systems

1. Introduction

Wildlife–vehicle collisions (WVCs) have progressively increased in both developed and developing economies. Every year, the United States (U.S.) registers 725,000 to 1,500,000 WVC cases, with over 200 humans dying annually. In comparison, Canada records a wildlife roadkill rate of 4 to 8 animals per day [1]. WVCs lead to severe economic losses, with the Ministry of Transportation of Canada spending over 600,000 CAD while cleaning up the collision sites. The U.S. Department of Transportation registered 200,000 WVC cases in 1990 and more than 300,000 in 2004 [2]. Generally, from 1975 to the mid-2000s, there were escalating deaths caused by collisions with cars in the U.S. Excitingly, the death trend has leveled off over the past decade. These deaths increased from 89 in 1975 to 223 in 2007 and declined to 185 in 2019. In 2019, the highest number of WVC fatalities occurred only during July–September [3]. Globally, amphibians and reptiles are the most endangered species, with records showing WVC cases of 180 amphibians and 72 reptiles on tertiary roads from March 2014 to October 2015 in a rural region in eastern Austria [4]. In 2015, the greater Mapungubwe transfrontier conservation area in South Africa recorded 991 and 36 roadkill on paved and unpaved roads, respectively [5]. The latter study shows



Citation: Nandutu, I.; Atemkeng, M.; Okouma, P. Intelligent Systems Using Sensors and/or Machine Learning to Mitigate Wildlife–Vehicle Collisions: A Review, Challenges, and New Perspectives. *Sensors* **2022**, *22*, 2478. https://doi.org/10.3390/s22072478

Academic Editors: Paul Fergus, Carl Chalmers and Serge Wich

Received: 25 December 2021 Accepted: 4 March 2022 Published: 23 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). that unpaved roads recorded less WVCs than paved roads. High numbers of WVCs were both in hot and wet seasons, while cold and dry seasons experienced the lowest numbers. In February 2012, researchers recorded 470 animal mortality rates due to WVCs in just two fortnights in South Africa. Birds had the highest numbers, followed by reptiles, mammals, and amphibians [6]. Roadkill and WVCs also happen in other parts of Africa, such as Tanzania [7,8]. In 2015, John Kioki et al. [7] recorded a death rate of 3% for domestic animals and 97% for wildlife due to WVCs. Their research findings show that roads are a potential threat to wildlife in East Africa.

The WVCs have caused death to humans and animals as well as damage to properties and roads, contributing to environmental conditions [9]. These collisions have affected the safety of humans, property, and wildlife. Despite the escalating numbers of WVCs, several methods adopted help solve the problem of roadkill. These methods include traditional (e.g., fencing, underpasses, and overpasses), intelligent systems (e.g., area cover, buried cable, autonomous vehicles, driver assistants, and break-the-beam), and machine learning algorithms. Some of these intelligent systems do not integrate machine learning algorithms to mitigate WVCs. Nevertheless, these intelligent systems and machine learning algorithms experience false detections when engaged in many accidents with animals on highways. By definition, machine learning is a sub-field of computational intelligence that exploits mathematical tools to make computers function like the human brain. It has gained popularity to address problems that are challenging to tackle by traditional computer science. Machine learning algorithms include but are not limited to support vector machines [10], naive Bayesian [11], and artificial neural networks (ANNs) [12]. The latter algorithm can minimize prediction and detection errors via the gradient descent based on the back-propagation algorithm. This capability of ANNs enables them to easily solve complex problems in different fields. A type of ANN is the feedforward neural network, which features an extraction process that is oriented from the input layers to the output layers, passing through the hidden layers. Moreover, another type of ANN is the recurrent neural network (RNN) [13], long short-term memory (LSTM), and convolutional neural networks (CNN [14]). Long short-term memory can mimic the behavior of randomly generated data over time. The LSTM does this via massive units with various components such as weights, activation functions, and backward and forward neurons connections to handle the vanishing gradient problem, which a very deep CNN would likely fail to address [15].

1.1. Problematic

Despite the escalating damage and danger that comes with WVCs, researchers engage in developing engineering solutions to monitor animals and mitigate the WVCs. Traditional approaches are implemented, such as wildlife crossings (e.g., overpass and underpass structures) and the installation of fences together with segments on highways. These traditional approaches are limited and are not always possible with steep rocky slopes and deep snowpack, among others [16]. Because of these limitations, other solutions such as wildlife-vehicle detection systems have received much attention. These include systems such as break-the-beam [17-21], area cover [17,18,22-27], buried cable [28-30], driver assistance [31], autonomous vehicles [32], and mobile mapping animal detection systems (ADSs) [33,34], as well as roadkill detection [35]. These ADSs are combined with traditional methods to enhance the prediction and detection of animals on the road. They also help alert the drivers to be aware of animal presence and the exact location of animals on the road. Although recent research on ADSs in combination with machine learning has emerged, it is more popular in autonomous cars, followed by driver assistants. In other systems which try to integrate machine learning and ADSs, their research is still in early stages using simulation techniques. The simulation process uses synthetic images with small training and test datasets resulting in low prediction accuracy [19].

Nevertheless, current survey and review studies on mitigating WVCs with machine learning algorithms and/or intelligent systems are non-exhaustive. In 2021, the Said

et al. [36] review on ADSs and characteristics of WVCs is vital, and in 2020, Raphaela Pagany [37] discussed WVC influencing factors. However, their works did not focus on machine learning methods that prevent WVCs and current datasets used to mitigate WVCs. Because of these gaps, the motivation for this paper is six-fold:

- Understand and critique the current ADSs used to mitigate WVCs.
- Explore the negative factors contributing to WVCs; these negative factors should act as features when developing intelligent systems that prevent WVCs.
- Discuss and criticize the current systems that integrate machine learning to mitigate WVCs.
- Identify the challenges and the gaps for the reviewed current systems that prevent WVCs and identify potential solutions or new perspectives.
- Propose future research directions.
- Discuss the applicability and feasibility of the proposed solutions.

This investigation will fill the research gaps by comparing these technologies in terms of their strengths, weaknesses, limitations, and the way forward in the field.

1.2. Contributions

We analyze negative factors that influence WVCs, such as human behaviors, animal behaviors, road features, and climatic changes. These factors are key features to consider when developing ADSs and machine learning to prevent WVCs. We review existing systems that prevent WVCs, e.g., area cover systems, break-the-beam systems, and other ADSs. We discuss area cover systems classifications with sensor types such as active infrared sensors, passive infrared sensors, passive video sensors, and active microwave radio sensors. Besides classifying with sensor type, we state the strengths and weaknesses of each area cover system. In break-the-beam systems, we use microwave radio, infrared, and laser sensor types. In addition, we discuss the strengths and weaknesses in each study. We also discuss the strength and weaknesses of the buried cable, mobile mapping, driver assistance, autonomous vehicles, and roadkill ADSs. We conduct discussions on different types of machine learning that prevent WVCs. Supervised methods such as support vector machine (SVM) [38], maximum entropy environmental niche modeling [39], CNN [34,38,40], YOLOv3 [41], Xception [34], VGG16 [34], VGG19 [34], ResNet50 [34], inception [34], and unsupervised methods such as K-nearest neighbors (KNN) [19], PCA [38], feature extraction [19], frame differencing [19], template matching [19], LDA [38], LBPH [38], as well as ensemble learning such as random forest [19] have achieved promising results, among others. We find a non-exhaustive work in the literature that explores self-supervised and semi-supervised machine learning algorithms and how they contribute to mitigating WVCs. Equally, we discuss state-of-the-art datasets that could help improve model accuracy and minimize WVCs. Additionally, we record issues and challenges arising from the review and suggest a way forward. Furthermore, we explain the applicability and feasibility of the proposed solutions to existing damaged roads, accident car collisions, humans, animals, and the environment. Finally, we design a continuous product development lifecycle to help engineering teams determine the feasibility of the proposed solutions.

1.3. Manuscript Organization

The organization of this work is as follows: Section 2 discusses the methodology adopted in this work. Section 3 discusses four factors contributing to human–wildlife conflict and eventually WVCs. Section 4 reviews existing intelligent systems that prevent WVCs. These are area cover systems, break-the-beam systems, driver assistants, autonomous vehicles, roadkill detection, mobile mapping systems, as well as buried cable ADSs. Section 5 shows a collection of the datasets used to mitigate WVCs. Section 6 conducts a survey of machine learning methods and how they prevent WVCs. Section 7 identifies issues from the review and proposes potential future research directions. The same section also discuss the applicability and feasibility of the proposed solutions.

2. Methodology

To define the methodology supporting this review, we recall at least three dominant approaches in literature reviews: the narrative review, the systematic literature review, and the bibliometric analysis.

The narrative literature review typically provides a broad overview of a specific topic chosen by the author. It is based on the literature available about the designed topic. It is often descriptive and written in an easily readable format [42,43]. The systematic literature review has two objectives: first, perform an extensive literature search through a very detailed process and then critically assess the selected literature [42,43]. Last but not least, the bibliometric analysis aims at developing a quantitative investigation and relies on statistical methods to evaluate several characteristics of specific bibliographic information such as journals, research institutions, geographic location, and other quantifiable characteristics [44].

In the present work, we conduct a combination of a systematic review with some elements of bibliometric analysis.

2.1. Formulating Review Questions

The study is guided by formulating a comprehensive list of research questions. The questions guided reviewers on the focus, scope, applicability, and feasibility of the research. We define the study's goal to draft the review questions easily. We aim at reviewing the literature on intelligence systems using sensor technology and/or machine learning to mitigate WVCs. We focus on reviewing datasets that help mitigate WVCs. Additionally, we define a list of issues and challenges from the review and explain the applicability and feasibility of the proposed solutions. To achieve the aim, the authors formulate and implement research questions in Table 1 to guide them in writing this manuscript. The questions help the authors define the scope of the review and set out what they wanted to answer.

RQNi	Question	Description
RQN1	What are the negative factors that lead to the occurrence of WVCs?	To identify negative factors that contribute to human–wildlife conflicts and WVCs.
RQN2	What ADSs are deployed to mitigate WVCs in the primary studies?	To identify all the ADSs and techniques used in the selected primary studies. State the results, strengths, and weaknesses of each system.
RQN3	What types of datasets are currently used to mitigate WVCs?	The different types of datasets, e.g., text, images, time series, etc., and what machine learning tasks can use the data identified.
RQN4	What types of machine learning algorithms are used to mitigate WVCs?	To identify all the machine learning algorithms used in the selected primary studies to mitigate WVCs.
RQN5	What are the limitations of the primary studies in mitigating WVCs?	To identify all the constraints or weaknesses of the ADSs and machine learning methods in mitigating WVCs and propose future recommendations.
RQN6	How applicable and feasible are the proposed solutions?	To show how the solutions minimize the issues and challenges arising from the review.

Table 1. Research questions and their descriptions. RQNi represents the research question number i.

2.2. Search Databases

The search databases for this study are Scopus and Google Scholar. The articles are peer-reviewed (PR) and non-peer-reviewed (NPR). We further search specific websites that

had relevant information. Scopus is an international database of PR publications from all over the world [45]. Using Google Scholar, an advanced search engine, helps to cover citations that are not in Scopus [46]. We search each database and analyze results based on the search string.

2.3. Locating Studies through Searching

To identify relevant publications, we perform a bibliographic search both in Scopus and Google Scholar during April 2020 and November 2021. We also extract content from specific websites which had reliable information. We extract and review a total of 136 manuscripts in this study. To extract manuscripts to review, we use search terms and words such as "animal-detection systems", "break-the-beam", "area-cover", "mobile-mapping", "buried cable", "driver assistance", "unmanned aerial vehicles (UAVs)", "drone". To specifically retrieve only manuscripts associated to ADSs that help mitigate WVCs, and those that publish datasets to help reduce WVCs, the term "datasets + wildlife-vehicle collisions" is considered. Detailed general terms and terms for each instance are shown in Table 2. Further, the words "machine learning methods + mitigation + wildlife-vehicle collisions" helped retrieve only manuscripts that use machine learning methods to mitigate WVCs.

Table 2. Search terms used when retrieving manuscr	ripts from the different internet sources
--	---

General Term	Term
break-the-beam + animal-detection systems	break-the-beam/animal-detection systems/mitigation/roadkill/animal-vehicle collisions and wildlife-vehicle collisions
area-cover + animal-detection systems	area-cover/animal-detection systems/roadkill/mitigation and/animal-vehicle collisions and wildlife-vehicle collisions
mobile-mapping + animal-detection systems	mobile-mapping/roadkill/animal-detection systems/mitigation/animal-vehicle collisions and wildlife-vehicle collisions
buried cable + animal-detection systems	buried cable/roadkill/animal-detection systems/mitigation/animal-vehicle collisions and wildlife-vehicle collisions
driver assistance + animal-detection systems	driver assistance/roadkill/animal-detection systems/mitigation/animal-vehicle collisions and wildlife-vehicle collisions
unmanned aerial vehicles (UAVs) + animal-detection systems	unmanned aerial vehicles (UAVs)/animal-detection systems/roadkill/mitigation/animal-vehicle collisions and wildlife-vehicle collisions
drone + animal-detection systems	drone/animal-detection systems/roadkill/mitigation/animal-vehicle collisions and wildlife-vehicle collisions
machine learning methods + animal-detection systems	machine learning methods/animal-detection systems/mitigation/roadkill/animal-vehicle collisions and wildlife-vehicle collisions
datasets + wildlife-vehicle collisions	datasets/wildlife-vehicle collisions/animal-detection systems/mitigation/roadkill/animal-vehicle collisions and wildlife-vehicle collisions

2.4. Inclusion and Exclusion

Our methodology identifies eligible articles we obtain from Scopus and Google Scholar datasets. We compare manuscripts and remove duplicates. Later, we screen the remaining manuscripts for inclusion and exclusion based on defined criteria; this process is shown in Figure 1. We also obtain additional information from websites. Information from websites and Google Scholar also contain company reports.

For any manuscript included in this review, we consider studies that:

- Refer to the use of ADSs and machine learning methods that mitigate WVCs.
- We discuss datasets used to mitigate WVCs.

We exclude studies that:

- Focus on ADSs or machine learning that count and monitor species minus mitigating WVCs.
- We did not consider any machine learning algorithms that detect hotspots.
- Articles that did not satisfy the search criteria.
- Review papers.



Figure 1. Flow diagram of exclusion and inclusion process.

2.5. Data Extraction and Analysis

We consider a total of 56 and 53 manuscripts from Scopus and Google Scholar, respectively. Additionally, the authors collect manuscripts from government websites (gov't), conservation websites (cs), education websites (educ), and news websites (news) with a total of 5, 7, 11, and 4 manuscripts, respectively, totaling to 27 additional manuscripts. Although we find these manuscripts suitable for review, they did not appear in both Scopus and Google Scholar databases. The 27 manuscripts have clear objectives and accurate information which prompted the authors to consider them for review. We summarize and analyze these manuscripts in Figures 2–5.

Figure 2a visualizes the total number of manuscripts per each internet source. We extract the highest number of manuscripts from Scopus followed by Google Scholar, education websites (website (educ)), conversational websites (website (cs)), government websites (website (gov't)), and finally from news websites (website (news)). Figure 2b visualizes the different document types reviewed with the associated number of manuscripts. The latter figure ranks articles as the highly reviewed manuscripts in this review followed by reports, conference papers, then websites and books. It is worthy to note that these document types contain both PR and NPR manuscripts.

Figure 3a shows the distribution of years captured in this review. For each year, we show the total number of manuscripts. The coverage of the manuscripts in this review range

from 1975 to 2021, with some years seen as inactive. We observe that research in this field has been neglected, with fluctuations in the yearly number of papers published from 1998 to 2021. The highest number of papers published per year are 14 manuscripts. This shows that more research is required in this field in mitigating WVCs using intelligent systems and machine learning methods. Even though the number of manuscripts progressively published are few, the authors majorly archive them in PR journals as shown in Figure 3b. Figure 3b shows PR and NPR manuscripts. PR manuscripts are many compared to NPR manuscripts. Both PR and NPR manuscripts may contain articles, reports, conference papers, and books; document types are presented in Figure 3b.



Figure 2. (a) Displays the total number of manuscripts extracted from each internet source, and (b) visualizes the total number of manuscripts associated to each document type.



Figure 3. (a) Counts manuscripts published per year, and (b) summarizes review status of manuscripts, i.e., NPR and PR manuscripts, respectively.

To evaluate the relevancy of the research conducted and databases considered, we extract the number of citations as shown in Figure 4. In Figure 4a, we visualize the number of citations per each manuscript associated to a specific database. We observe that manuscripts we extract from the Google Scholar database are cited highly compared to the Scopus database. Google Scholar, Scopus, website (educ), website (gov't), website (cs), website (news) have 310,593, 42,880, 839, 299, 2, and 0 citations, respectively. The content

from websites is composed of company reports. Google Scholar is cited more than the Scopus database due to the following influencing factors:

- Scopus contains only PR manuscripts from Scopus indexed journals.
- Google Scholar contains both PR and NPR manuscripts such as technical which we consider in this review.
- Scopus citation count includes only the number of times the publication was cited by articles from journals that Scopus covers.
- Google Scholar counts citations from every journal published; it may include document types such as books, conference proceedings, dissertations/thesis, patents, technical reports, or other types of publications.

The influencing factors justify the high number of citations witnessed for manuscripts extracted from the Google Scholar database. Based on this, we find the Scopus database very relevant. Figure 4b shows document types with associated number of citations. The number of citations from each document type show that conference papers are highly cited with 151,318 citations followed by journal articles with 117,646 citations, books with 85,106 citations, reports with 516 citations, and 21 citations recorded for websites; citations from websites are possible because websites contain company reports that we include in this study.



Figure 4. (a) Displays the total number of citations per database, and (b) visualizes the number of manuscript citations per document type.

Using the same principle, Figure 5 also shows the relevancy of the manuscripts when correlated to the number of citations. Figure 5a visualizes the number of citations of each PR and NPR manuscript. The PR manuscripts are highly cited compared to NPR manuscripts. Figure 5 shows the number of manuscripts per each total number of papers per year. In 2016, the published manuscripts are highly cited, followed by 2014, 1980, 2017, 1985, among others. NPR manuscripts equally have many citations showing that the research published is vital to the domain. The NPR manuscripts have 70,251 citations, while PR manuscripts have 284,362 citations.

Due to the non-exhaustive research conducted on ADSs and machine learning methods that mitigate WVCs, we consider a broad scope from 1975 to 2021 as displayed in Figure 3a. Figure 3a justifies that there is no exhaustive systematic research conducted in this area since there are fluctuations in publications in the different years, with many inactive years showing no research published in the domain. The number of citations in Figure 5b for each manuscript published each year equally fluctuates, with 2016 having the highest number of citations. Due to these unidentified research publication trends, researchers should engage in publishing research that mitigates WVCs.



Figure 5. (a) Visualizes manuscript citations per review status showing NPR and PR manuscripts, and (b) shows relationships between the number of citations per each manuscript and year of publication.

3. RQN1: What Are the Negative Factors That Lead to the Occurrence of WVCs?

Globally, wildlife accidents with cars pose a safety threat to both animals and humans. In this section, we explore the factors influencing the occurrence of these human–wildlife accidents plus attacks. The influencing factors include human behaviors, animal behaviors, road features, and changes in climatic conditions. These factors warrant a detailed description and understanding as they are crucial features to consider during the implementation of intelligent systems that aim to prevent WVCs.

3.1. Road Features

A road is a path over which vehicles and other traffic may lawfully pass. It includes pathways and may also include culverts, bridges, and land required for future widening [47]. Roads are characterized by curve road sections, no road marking, smooth, rut, corrugations, express roads, primary roads, secondary roads, double lane line marking, and rural location [48]. In road engineering and construction, the classification of a good road is by its shape, width, thickness, and the materials it is made of [49]. Materials of the road are usually found on a railway crossing, cobbled road, pothole, damaged road, speed bump, manhole covers, and a road in good condition [50]. Roads are a barrier to animal movement as they contribute to roadkill during crossing attempts or behavioral avoidance. WVCs have negative consequences, such as demographic and genetic, that can ultimately result in local or regional extinction [51]. Possibility of extinction is minimized by implementing road structures such as fences to avoid animals from moving toward the roads [52]. The fences limit the movement of animals and therefore reduce roadkill and mortality rates. However, some animals can still jump through these constructed fences and attempt to cross the roads. Road sections with high rates of WVCs are associated with areas with high habitat diversity, limited crop cover, and fewer number of structures [53]. Research shows that there is a significant decline in WVCs when road features such as fencing and pavement are adopted [54]. Collinson et al. [55] designed a low-level roadside fence to direct animals to underpasses as a way of mitigating WVCs of small vertebrates. The study was not satisfactory due to low witnesses of dead animals detected after the installed barrier sites. This study justifies that traditional structures do not totally eliminate WVCs, and the domain requires more research to identify traditional structures or intelligent systems that can mitigate WVCs with extremely high accuracy.

In another work, records of WVC hotspots are more on highways for mammals and birds and riverside roads for amphibians and reptiles [56]. WVCs are more likely to occur alongside roads with thick vegetation cover or habitats which hinder driver visibility [7]. In 2013, Seiler [57] emphasized that the mortality of these animals increases due to high traffic speeds and volume. Road traffic accidents kill hundreds to millions of animals every year, posing a critical threat to the populations of many species [58]. World Health

Organization (WHO) states that over 90% of the world road fatalities occur in low-income and middle-income countries, even though these countries have only about half the world's vehicles. Driver injuries increase due to factors such as lateral crosstown roads, less traffic volume, increase in the percentage of heavy vehicles, wider lanes, lack of road markings, and violation of driving laws and regulations [59]. Road safety is one of the considerations in support of the 2030 agenda for sustainable development [60]. As a mode of contributing to sustainable development goals, the United Nations conference on trade and development wrote a report on road safety, contributing toward the WHO road safety agendas. The document records show that more than 3500 road deaths occur worldwide every day and 1.25 million each year, with the addition of up to 50 million persons facing injuries in road accidents [61]. Furthermore, the annual road traffic predicts that deaths are to increase to around 1.9 million by 2030 and become the seventh leading cause of death [62]. The draft report on road safety states that increase in mortality rates is due to overspeeding, drinking and driving, and lack of motorcycle helmets and seat belts. To address this problem of WVCs, some types of road mitigation measures are available that aim to reduce wildlife mortality on roads. Some of these measures are traditional methods, e.g., overpasses, fences, and underpasses that prevent animals from accessing roads—these traditional solutions minimize the death rate of wildlife. For an effective and reliable approach to avoid WVCs, we propose adopting traditional methods together with ADSs and further integrate them with machine learning algorithms as a mitigation strategy since it contributes to reducing mortality rates [52]. In this review, we give a comprehensive list of proposed solutions for road safety in Section 7.2.

3.2. Climate Change and Season

Climate change inconsistencies are a threat globally and are now slowly affecting Africa [63,64]. These changes in seasons are reactions of the greenhouse effect by greenhouse gases. Greenhouse gases include carbon dioxide, methane, dinitrogen monoxide or nitrous oxide, water, perfluorocarbons, and sulfur hexafluoride. As well, halocarbon gases are composed of trichlorofluorocarbon and dichloroflurocarbon. These atmospheric concentrations continue to be possible due to anthropogenic activities (human activities) that hinder the balance of atmospheric gases [65]. The climatic change also creates different weather variations and seasons such as dry (drought) and wet (food) conditions, number of hurricanes, and temperatures [66]. These weather alterations affect bird life and animal life with their behaviors, foraging, migration, growth, and reproduction patterns. When the weather variations affect bird and animal lives, bird and animal movement patterns fluctuate, leading to variations of WVCs on highways. Besides, global warming contributes to a shift in seasons. The variations in seasons contribute to species extinctions since many of the species live in areas that are severely affected by climate change, and climate change is happening too quickly for many species to adapt [67]. Lately, global warming is intriguing. In addition, it has led to the global decline in wildlife, leading to wildlife effects such as the loss of habitats, damage to oceans, coastlines, and coral reefs due to sea-level rise, and increased storm activity, as well as more frequent and intense periods of drought [68].

Climate change and seasons alter wildlife movement and distributions due to the change in position and location of habitats, potentially leading to associated location changes in WVCs. The proposed development of movable structures can be a solution other than overpasses and underpasses. Besides, it is transferable from one location to adapt to the changing weather conditions that affect the animals [69]. In 2012, the Food and Agriculture Organization of the United Nations stated the consequences of climate change. The adverse effects are: (1) The changes in ecosystem and landscape, which reduces biodiversity, leading to significant changes in disturbance patterns, e.g., more fires, more droughts, and more floods, as well as loss of species due to mistiming, competition, and stress from new species within an ecosystem. (2) The variations in species distribution due to temperature, rainfall, and geographical barriers. It is important to note that natural and human-made obstacles to movement are a problem to many animals as they try to

move. These animals move to search for new conducive habitats in response to changing conditions. (3) The human–wildlife–livestock conflicts due to increasing human population densities and encroachment of human settlements and activities into wildlife habitats contribute to rampant conflicts between humans and animals. These are majorly common in rural areas where humans depend more on agriculture for farming. These conflicts pressure animals since their habitats have been encroached on. (4) Wildland fires caused by climate change. These affect slow-moving animals at risk of mortality from flames and smoke. Other animals change their habitats due to lack of food, competition for the same territories, or accessing shelter and death due to starvation, and (5) wildlife health, diseases, invasive species, and pests.

The impact of climate change is rampant and includes permanent changes in physical conditions. Examples of these changes are snow cover, permafrost, sea level, and extreme increases in the irregularity and severity of extreme weather events such as droughts, floods, and storms [70]. The impact forces animals to loiter away from their habitats in search of a conducive environment such as green pastures. This act leads them to highways or crossroads. Animals found on the highways usually collide with vehicles leading to WVCs, which increase rampantly. In 2017, in Brazil, the study evaluates variables such as seasonality, climatic (mean temperature, relative air humidity, and accumulated precipitation), and average daily traffic on vertebrate roadkill rates [56]. The highest numbers of roadkill records in the rainy season are reptiles. Roadkill was high for birds during summer, mammals increase from spring to fall, and during fall, the roadkill for amphibians and reptiles was at its peak [71].

3.3. Human Behaviors

The behaviors of humans are a threat to conservation in wild reserves. As such, contributing to habitat loss through deforestation, as well as poaching, wildfires, conurbation, and climatic changes in the earth's atmosphere. The climatic changes are a result of the released amounts of greenhouse gases and aerosols (small particles), burning fossil fuels that release carbon dioxide into the atmosphere. Because of this damage to the atmosphere that humans cause through anthropogenic activities, most blame for climate and season alterations is partially due to their behaviors. Humans possess different behaviors from person to person. Each human has different abilities, aspirations, motives, perceptions, values, attitudes, and personality. Concepts such as actions, activities, and behaviors are essential in understanding human behavior [5]. Actions are short and visible movements such as taking a key, opening the door of a vehicle, and grasping. These actions can also be invisible such as perceptions, attitudes, thoughts, feelings, and physiological processes. On the other hand, activities are composed of several actions such as cutting down trees, poaching, and driving a car. When we combine activities and perform them at different time intervals, we refer to them as behaviors. Behaviors are those such as overspeeding, frequent braking, and poaching, among others [9]. It is worthy to note that actions and activities all contribute to WVCs on highways.

The mentioned activities and actions are driven by the underlying cognitive, emotional, and physiological processes [72]. Social psychologists believe that attitudes contribute to human behavior theory, e.g., what people think and do. They rely majorly on the attitude-based questionnaire by Ajzen and Fishbein [73]. This attitude-based questionnaire has four components that eliminate the difficulty in recognizing the attitude-behavior relationship. These components are beliefs, attitudes, intentions, and behaviors.

Research contains majorly two cognitive theories, which are the theory of reasoned action (TRA) [73,74] and its extension, the theory of planned behavior (TPB) [75]. These theories offer conceptual frameworks for understanding human behavior. In TRA theory, behavior intention combines a person's attitudes and subjective norms. TPB theory is an extension of TRA theory. TPB states that behavior intentions are usually determined by a combination of a person's attitudes, subjective norms, and perceived behavioral control. The TRA theory states that a person's intention to perform or not to perform a

behavior at a specific time and place, as well as the more the person thinks other people (friends, family, or society) would love them to perform that behavior (subjective norms), are the immediate determinant of that action, i.e., intentions (behavior intentions) to perform the behavior. This TRA theory is composed to help predict and explain volitional behavior. TRA components are the determinants of intentions, such as attitudes toward behavior determined by beliefs about the behavior and subjective norms. Psychologists define attitude as a tendency of an individual to evaluate an entity such as a person, place, behavior, or thing. The evaluation performed should be with a degree of favor or disfavor [76]. Attitudes are composed of a combination of behavioral beliefs (strong beliefs regarding behavioral outcome) and outcome evaluation (evaluation of the advantages and disadvantages of behavior outcome) [77]. Psychologists emphasize that attitude toward an object is not related to the attitude to behavior toward that object. For example, someone may have a favorable attitude toward a rhino (object). However, they will go ahead and engage in poaching (behavior toward objects) simply because they want the rhino horn, believed to be medicinal. A subjective norm is what we think other people will think of us if we perform (or do not perform) the behavior. Subjective norms are composed of normative beliefs (strong beliefs regarding how important people will agree or disagree with the behavior) and motivation to comply (motivation to comply with important people) [77]. Subjective norms are a result of social and environmental surroundings. Social norms and taboos play a significant role in determining appropriate behavior in society and the motivation to adopt the behavior. In 1988, while conducting a meta-analysis, Sheppard et al. [78] showed that TRA is a powerful predictor of behavior. In general, Ajzen et al. [74] highlight that the more positive a person regards attitude and subjective norm as important, the more likely they are to form intentions to engage in the behavior.

The TPB theory focuses on perceived behavioral control; it looks at a person's belief in self-control over their behavior. TPB's key component is perceived behavior control. Perceived behavioral control refers to a person's perception of how easy or difficult it is to perform any behavior of interest to them. Perceived behavioral control varies across situations and actions, which results in a person having varying perceptions of behavioral control depending on the situation. Perceived behavior control is composed of control beliefs and power beliefs [77]. Control beliefs are strong beliefs about what allows or avoids the performance of the behavior. Power beliefs are power perceived presence of factors that may limit or enhance the performance of a behavior. Human decision-making traits are an attitude of a person toward the behavior, subjective norms regarding behaviors, and a person's perceived control over the behavior [79]. Generally, a positive attitude and positive subjective norms result in greater perceived control. These increase the likelihood of intentions governing changes in behavior. In other social research, there is an alternative approach that predicts human behavior. The approach looks at behavior aspects of the lives of people as the basis of the question design [80]. This new approach emphasizes replacing attitude questions with questions about the respondent's environment, consciousness, knowledge, and behavior [81]. Drivers and other stakeholders collide with vehicles due to the power that lies within the two models, TRA and TPB. Their actions and activities are highly attributed to human behaviors. In Table 3, the strengths and limitations of the TRA and TPB models are highlighted.

Table 3. Strengths and	limitations of cogni-	tive theories: T	FRA and TPI	B models
------------------------	-----------------------	------------------	-------------	----------

Model	Strengths	Limitations
TRA	 Subjective norms and attitudes toward predicting behavior were important Future behaviors were determined by behavioral intentions 	 The theory focuses on behavior under volitional control The model is not used due to the theory of planned behavior
TPB	 The introduction of perceived behavioral control covers the non-volitional behaviors that were eliminated in the theory of reasoned action Gaston et al. [82] highlighted the theory as good at explaining intentions 	• Intention prediction is more than behavior prediction. This is due to temporal element [75]

3.4. Animal Behavior

Animal behavior refers to ways in which animals associate with other organisms and the physical environment [72]. On the other hand, behavior is the change in the activity of an animal in response to an external or internal cue [83]. The field of animal behavior is composed of research on: (1) feeding behavior [84], (2) habitat selection [85], (3) mating behavior [86], and (4) social organizations [87]. Animals classified according to their feeding behavior are, e.g., herbivores, carnivores, omnivores, scavengers, and vegetarians. In 1986, behavior evolution was based on assumptions that specific genes hardwire innate behavior. As well, learned behavior is attained by a more adaptive nervous system than innate behavior [88]. These assumptions lead to two categories of behaviors which are innate while others are learned [89]. The categories are determinants of how animals behave the way they do. The physiology and anatomy of an animal also contribute to behavior—both external stimuli (threats from other animals, sounds, smells, weather) and internal stimuli (hunger, fear) prompt behaviors [90]. Both innate and learned behaviors lead to WVCs as animals make decisions to frequently cross the roads.

In wildlife conservation, harsh living conditions, e.g., drought, destabilize animals tempting them to move toward roads [91]. Animal movements are disadvantageous and lead to a lot of negative results, e.g., (1) a decline in biodiversity (due to loss of habitats), (2) the introduction of exotic species (foreign, non-indigenous, or non-native species), (3) over-harvesting of biodiversity resources (illegal hunting, cutting down trees), as well as homogenization of species in agriculture (species lose their "character" through particle size reduction in cells in animals or plants) [92]. WVCs lead to a decline in diversity of animal species in wild reserves. The movement patterns of different animal species toward particular roads continue to expose many repertoires of movement patterns that vary in duration and complexity [93] due to animal behaviors [94].

To enable conservationists and researchers to understand animal behavior easily, implementation of animal tracking technology in game reserves has been ongoing and helps to capture and extract behavioral data [95], using monitoring tools such as GPS and VHT [96,97]. In South Africa, behavioral data from tracking systems, such as animals' daily movements, behavior, and diet, are collected from collars fitted on animals [98,99]. Classification of some animal species behavior types, such as walking, eating, foraging, lying, standing, among others, are observed using GPS tracking and has scored an accuracy of over 80% [100]. The latter research shows that accelerometer data plus GPS tags contribute to a promising result in distinguishing more behaviors than could be classified with GPS alone. Adopting machine learning algorithms and integrating it into tracking systems to classify animal species behavior types while understanding brain-behavior relationships achieves acceptable accuracy. Their implementations using CNNs seek to understand five different behavior tasks in mice and humans through models that generalized to different behavior tests [101]. We note that both supervised and unsupervised machine learning methods help understand or monitor animal behavior. Some examples of this application are the prediction of animal behaviors from accelerometry data in pigs [102] and the classification regression decision trees used to track cows in open and habitat fields using differences in movement metrics between behavior types from GPS data [103]. In the latter, the GPS data with oneminute time step interval from the open field scored an accuracy of 70%, while data with 12 and 2 min time step interval failed to classify. These results show that the time step interval should be short enough for the behavior to be defined by its characteristic movement metrics. Moreover, data obtained from forested areas scored higher accuracy compared to an open field. Unsupervised algorithms are used extensively in identifying and modeling biologging and telemetry data such as hidden Markov models or Gaussian mixture models [104].

In summary, we have discussed the prevailing conditions that contribute to WVCs. In Section 3.1, we understand the contributions of roads and road features in reducing roadkill. Section 3.2 discusses how alterations in climate change and season boost animal mortality rates due to WVCs. Sections 3.3 and 3.4 discuss the behaviors of humans and

animals, respectively. To minimize these WVCs, we introduce Section 3 and discuss existing ADSs that prevent WVCs.

4. RQN2: What ADSs Are Deployed to Mitigate WVCs in the Primary Studies?

The existing intelligent technological systems used today to prevent WVCs are areacover, break-the-beam, mobile mapping, buried cable, driver assistant, and autonomous vehicles ADSs. These operate independently, and some use machine learning algorithms to mitigate WVCs. Area coverage sensors and break-the-beam sensors are the two frequently used technologies to detect large animals within the range of the sensor either at a specific angle or in a linear shape as animals approach the road. These two systems are either active or passive [16]. Mobile mapping systems and roadkill detection systems detect small animals on the road. Buried cable systems can detect both the crossing of largeand medium-sized animals while providing data on their location along the length of the cable. Driver assistants and autonomous vehicles are the latest technologies deployed with machine learning models to help detect small, medium, and large animals. Even though these two seem promising in the field, the implementations are not exhaustive in developed economies, with less to no attempts made in developing and under-developed economies.

4.1. Area Cover Systems

For animal detection on the road, an area-cover system uses passive video sensors, passive infrared sensors, and active microwave radio sensors [105]. Table 4 identifies and describes each sensor type used by the area cover systems in this study.

Sensor Type	Description
Active infrared sensors [106]	 Provides its own source of energy for illumination It has a transmitter which emits invisible active infrared beams and the receiver which receives and analyzes the beams Detects intruders passing between the detection area
Passive infrared sensors [107]	 This sensor works by detecting the energy emitted by other objects such as an animal This sensor uses the concept of black body radiation The objects emit heat in the form of infrared radiation in the detection area Any intruders in the detection area are detected
Passive video sensors [106]	 This technology detects and records objects and evaluates scenes recorded by a video camera
Active microwave radio sensors [106]	 This technology uses both passive and/or active remote sensing It emits microwave pulses and measures the reflection of a moving object They cover a larger area than infrared sensors Are vulnerable to electrical interference

Table 4. Description of sensors deployed in area cover systems.

An area cover system detects large animals in a distance of up to 50 meters within a 60° horizontal angle [108]. The system operates by reacting to changing physical conditions that interfere with its electrical properties. They capture stimuli such as heat, sound, and movements that animals induce. These stimuli convert to digital signals and are then sent to a computer for processing and analysis. Area cover systems are passive or active. Passive area cover systems are composed of passive infrared and video detection, and active area cover systems are composed of microwave radar. These systems have successfully provided animals the liberty to move compared to the traditional fencing systems. However, these systems still face excessive false positives under vegetation cover [109]. In 2009, researchers developed and evaluated a test-bed composed of five cone-shaped area cover systems. These systems detect animals based on motion and body heat in an area and range from the sensor. Their records show a decrease in false positives and an increase in false negatives in passive area-cover systems due to high winds [18].

In 2003, the development and deployment of the thermal animal detection system (TADS) was successful. The TADS solved the intense collisions between birds and wind turbines in Denmark. To quickly achieve a reliable system for the TADS, thermal cameras are associated with hardware and software. These system recordings show birds migrating

and nearing the wind turbine blades of the offshore wind farms. The system is sometimes effective under harsh conditions with poor visibility (fog, rain, or snow). A threshold temperature level is defined to ensure only sequences of birds observed and captured passing through the area cover system can eliminate false detections. The quality of the captured images are equally good in both compressed and non-compressed regimes. However, this work shows that collisions of turbines with birds are possible and highest at night when the visibility of the thermal cameras is poor [24].

The design and implementation of the flashing light animal sensing host (FLASH) in the U.S. helps detect deer on the U.S. Highway 30 pathway [25]. This system consisted of three data collection systems installed on this highway. These systems are the FLASH, warn, and motorist-speed systems. The FLASH system is composed of infrared sensors. These sensors trigger each time a deer moves through the detection area by emitting a signal to the receiving unit to activate flashing lights which are usually above permanently visible warning signs. The lights alert the drivers who are distant about the existence of a deer on the road. The intention is to prompt drivers to slow down the car's speed determined by the motorist-speed system embedded with sensors. Next, sensors gathered data on vehicles on the highway before encountering the warning signs on either the right or the left sides of the road. Whenever the vehicles approach the detection area, speed sensors gather speed data on vehicles after viewing the sign and are already crossing. Moreover, the sensors register the number of axles of individual vehicles that pass and distinguishes them between automobiles and semi tractor-trailers. Further, the experts capture the time and date when each vehicle crosses the sensor. On manipulations of speed-sensor data and the FLASH data, the researchers are able to determine vehicle speed, vehicle type, date, time, and the status of the FLASH system (activated or not) for each vehicle. The data from the FLASH system is stored and downloads enabled remotely via modem. Results show that each factor significantly affects speed reduction for semi tractor-trailers under normal system operations. Factors, e.g., date, had a probability value equal to 0.003. For the time of day, the probability value is less than 0.001, and treatment had a probability value of less than 0.001. Automobiles also had a significant speed reduction for each factor, e.g., time of day had a probability value less than 0.001 and treatment had a less than 0.001 probability value. The date had no significant impact on the automobile's speed because the date had a high probability value of 0.101. Some months influenced vehicle speed more than others. During December, semi tractor-trailers reduce their speed to 1.1 km per hour more than during January, February, and March. The vehicles reduce their speed to a minimum in the daytime and relatively high at night each day. This happens when the lights are either activated or deactivated. For experimental treatments, results show that all factors are significant with automobiles having a probability value of less than 0.001, and semi tractor-trailer records treatment and time of day as significant. Results regard date as not significant since it had a high probability value of 0.247. However, even when the system results are favorable, the system is unreliable for the U.S. Highway 30 pathway. This system can still operate well on roads with a huge percentage of local traffic than the U.S. Highway 30 pathway. In improving the effectiveness of the U.S. Highway 30 pathway at reducing deer-vehicle collisions, a few recommendations are suggested, such as the installation of a deer-proof fence as well as implementing an education program for the local motorists explaining how the system operates.

In 2006, the Oh DEER, Inc. (Wayland, MA, USA) company developed an area-cover system which detects white-tailed deer (large animal) within a certain range of a sensor [17]. These sensors transmit and receive microwave radio signals to detect large animal movements. The system stores detection data such as identification number, type of sensor, detection value, whether the flashing lights are turned on or not, battery voltage, solar panel voltage, and temperature in the box with equipment on a flashcard. The last three parameters are allowed for identifying potential problems. The data are downloadable onsite or remotely through a modem and land-based phone line. The remote location system configurations can enable time delay for a signal turn-off. In 2013, Mukherjee et al. [27]

developed an area cover system by integrating both passive infrared and short-range. and installed the system comfortably in terrain landscapes. Their system monitors animals within the virtual trap wire. Further, their system experiences high false alarms during the day, false detection, detection delay, and difficulty in detection during bad weather.

In 2017, a sensor composed of Doppler radar technology is used in detecting large animals in a detection area and alerting drivers through activated warning signs [23]. The focus is to achieve the reliability and effectiveness of the ADS. In evaluating how reliable radars are in detecting large animals, they adopted thermal cameras in monitoring animals at the area coverage. Each time doppler radar detected an image of a large animal, the thermal camera captures images every three seconds and saves them temporarily. To measure the effectiveness of the systems in reducing vehicle speed, they install speed radars to record the speed of individual vehicles. The radar records speed between the northbound and southbound lanes as vehicles approach the detection area. Evaluation of the reliability and effectiveness of the system is through conducting tests such as walk-through tests and seasonal reliability tests. Even though the researchers consider this system reliable and effective, it still registers false positives and negatives. In the same year, Vikhram et al. [26] developed an ADS in the farm areas to prevent human-wildlife conflicts. They used a passive infrared sensor to detect animals' movement by sending a signal off the controller. In addition, they also use ultrasonic sensors to generate a sound sent by a global system for mobile communications to alert the farmer and department through messages about the presence of an animal. Conducting system testing under bad weather was successful. Doppler radar is used again in the detection of animals and humans under vegetation cover in 2018 [22]. The Doppler radar has an HB100 sensor used in transmitting and receiving microwave radiation. If the strength of the reflected signal is weaker than the one initially sent, then the radar will invoke a detection. The Doppler radar is placed and moved in front of the different crops. In addition, a dog or person is placed behind the crops. Under this situation, the radar is able to detect these animals. Detection of humans is in five out of seven cases and a dog in two out of seven cases. Although doppler microwave radar detects animals, we need more research to minimize the disturbing factors on the measurements.

We summarize area cover systems in Table 5 while showing the different sensor types, performances, strengths, and weaknesses.

4.2. Break-the-Beam Systems

Break-the-beam systems use transmitters and receivers with infrared, laser, or microwave radio signals. When an animal is present within the system's beam, a warning sign is activated. We describe break-the-beam ADSs currently deployed around the globe showing their strengths and weakness.

In the last decades, the number of WVCs in North America was alarming. These came with negative effects, prompting researchers to develop a break-the-beam system to mitigate roadkill [17]. The system detects large animals as they approach the roads by activating warning signs to alert drivers about the existence of animals on the road or in a near distance. The sensor technologies and systems experts—a vendor company—conducts field tests. The designing of the break-the-beam system is by the vendor company. The system consists of a transmitter that sends modulated signals to the receiver. Suppose an animal breaks the paired sensor beam. In that case, activated warning signs trigger an alert to the driver about the existence of an animal on the highway which prompts the driver to reduce the speed. Increase in alerts leads to fewer or less severe carcasses with animals. They save the recordings of all detection at a master station. Detection events are broadcast in real-time using the ultra-high frequency radio system. The detections enabled on-site monitoring of the ADS operation using a portable data radio connected to a laptop computer. The study saves every change in beam status at every date and time. However, this system has drawbacks such as (1) covering longer distances, which requires many sensors, hence making the system expensive, and (2) the system requires more hardware resources on curves, slopes, and vegetation landscapes.

Weakness Reference Year Sensor/Signal Type Results Strengths ٠ The study achieves good visibility with thermal cameras during severe fog and snowfall We recognized water condensing within the • metal box as a potential problem Good quality of compressed and TADS can record migrating birds non-compressed captured images Collisions of birds and offshore wind farms are approaching the rotating blades of a Thermal infrared A threshold temperature level captures only expected to continue during periods of bad 2003 ٠ Desholm [24] turbine, even under conditions with poor sequences of birds visibility visibility • The study was on only eider duck, so limited Vibrations from turbines did not affect the data of other species video recording Time delays 20 s during remote controlling had no practical problem ٠ Detects a deer Captures details about the vehicle and its The FLASH detects deer moving through the speed The system is not very effective and needed Categorized vehicles such as automobiles crossing area station on the road with high traffic Infrared Minimizing of speed reductions is successful and semi tractor-trailers Poor usage of the system by drivers due to lack 2004 Gordon et al. [25] Speed Able to store and retrieve sensor data from when tested during different times of the day, of an educational program to train users of the months of the year, date, treatment, among the global system for mobile system on how to use it other factors communications ٠ Flashing warning lights activated to alert drivers Detects animals within a certain range of a Save the data for the detection to use it for Microwave radio Detection delay 2006 Huijser et al. [17] sensor decision making Liberty to wildlife movement than Passive infrared and video Detect large animals Experiences excessive false positives ٠ ٠ 2008 traditional fencing systems Active (microwave radar) Detect large animals Cannot detect small animals ٠ ٠ ٠ Detection of animals based on heat and Passive IR 2009 Large detection areas compared to motion False detections break-the-beam ٠

Table 5. A comparison of areas cover systems in terms of their sensor types, performances, strengths, and weakness.

U.S. Department of Transport [109] Huijser et al. [18] System is less sensitive during high winds ٠ Infrared systems operate only at night ٠ Able to monitor animals beyond the virtual ٠ Radar-based system is placed at lower . Higher false alarms during the day tripwire height from the ground than an infrared Prone to false detection due to rain Passive infrared Radars cover longer-range monitoring ٠ No precise alignment of transmitter-receiver Mukherjee et al. [27] 2013 . Possibility of an animal becoming stuck inside Short-range radar than infrared cameras pairs the fenced section or roadways Installations can be in ridges, gullies, and Radars use fewer sensors and poles to ٠ Delays in detection and sending alerts cover roadway than infrared rocky outcrop landscapes Longer monitoring of detected animals (radar) •

Reference	Year	Sensor/Signal Type	Results	Strengths	Weakness
Mukherjee et al. [27]	2013	• Radar	 Large animal detection (LAD) reliably detects large animals in covered area Activates flashing beacons to alert drivers Influence of driver behavior so that drivers can avoid a collision 	 LADs deliver high reliability by reducing false alarms in detecting animals Achieves a reliable and accurate tracking in a low power Cost-effective and environmentally friendly manner It is a modular system, which makes it easy to upgrade with less effort 	• Delays in detection
Vikhram et al. [26]	2017	• Infrared • Sound	 Passive infrared and ultrasonic sensor detect an animal in detection area and send an alert to the controller presence of the animal and send an input signal to the controller Audio voice recorder and playback plays the sound to divert the animal At night when a flashlight is on, an alert to the forest department and farmer about the presence of an animal is made Solar powers the system 	• The design system will not be dangerous to animal and human being, and it protects farm	• The system is not extensively tested, under bad weather (rain, snow, etc.)
Huijser et al. [23]	2017	 Doppler radar Thermal camera Sensor 	 Detects large animals (in 113 m) Radar detects a person, and the warning signs are on as long as the person was still in the detection area 	 The system can further be programmed to detect large animals in a much longer distance (400 m) The high location of the Doppler radar reduced the risk of theft, vandalism, or accidental damage and certain types of false alarms Warning signs are activated whenever the system detects an animal Drivers informed about the detection zone a distance away from the detection area Detection of one animal multiple times while it is still present in the detection area This technology distinguishes between large mammals and other objects such as vehicles 	 False positives: the radar reports the detection, yet there is no large animal False negatives: the radar did not report any detection, yet there is a large animal
Shapoval et al. [22]	2018	• Doppler radar HB100	• Radar is able to detect dog and human when placed behind the crops	 System is able to detect these animals under different gardener crops placed in front 	 Generally, the system did not perform well in detecting objects under thick crop cover and it achieves unfavorable results on p-value

Table	5.	Cont.
-------	----	-------

In 2009, an installation of a test-bed of four break-the-beam systems are successful in detecting large animals [18]. These systems consisted of infrared, laser, or microwave radio signals. Most of these systems had two sensors on a pole in the middle of the 91 m distance, with the sensors facing opposite directions. On evaluating the systems, the researchers achieve excellent visibility associated with fewer false positives. However, these systems also experienced challenges such as (1) high error rates due to high temperatures; this is due to the failure of infrared sensors failing to differentiate between hot air and body heat of animals, (2) terrains are not suitable for the installation of break-the-beam, and (3) high false detections due to thick vegetation cover. In California, along Highway 3, engineers installation of a microwave break-the-beam ADS was successful the next year. The system site is used as a test-bed to determine factors to consider when selecting an ADS, and it also shows the reliability tests of the system [20]. The test-bed is composed of an animal enclosure, space for many ADSs, and six infrared cameras with recording capabilities. The animal enclosure included shelter, water, area for feeding. This system detects animals, records the date and time of each detection, and further records animal movement within the enclosure. This work investigates the reliability of the system. The reliability test results show that the percentage of false positives is 0.007% (1 false positive over 140 valid detections), false negatives is 0.03% (4 false negatives over 140 valid detections), and 148 intrusions in the detection area, of which the study shows 144 detections, which led to 97% detections of intrusions in the detection area. Although these system results show high detection rates, it still experiences false negatives related to the shortest animal, "sheep", in the enclosure. As a technique of eliminating the false negatives, we proposed cutting the grass low in the enclosure.

In 2016, installations of a roadside animal detection system (RADS) in Florida [21] was successful in helping to reduce WVCs that are rampant. The system uses an infrared beam to initiate communication between transmitters and receivers. A paired combination of these transmitters and receivers communicate through sending and receiving signals. When an animal crosses between the sensors, the light beam sends back a weaker signal. Each time a sensor receives a weak signal, a flashing light turns on, warning the drivers that an animal is on the highway. The system performance result is average in detecting target species. The average success rates were between 10.7 and 66%, and average failure rates were between 34 and 89.3% of target animals not detected, resulting in false negatives. Generally, 90% classifications of RADS activations are false positives (lights flashing and animals not present). These activations cause a significant reduction in vehicle speed. The effectiveness of RADS is also tested in a controlled setting using a driving simulator. Driver simulation produced positive results such as (1) quick reductions in speed when exposed to picture-based RADS signs than word-based RADS signs, (2) early braking in response to observations of an animal in the road, (3) relatively low WVC cases, and (4) picture-based RADS signs create alertness of drivers more than word-based RADS signs. These tests show that the system is reliable and effective in detecting large animals and improving motorcyclists' road safety. However, the system still experiences downfalls such as false detections, malfunction, and drivers' ignorance of the usage.

In 2019, Antonio et al. [19] simulated an ADS that detects animals on the road and alerts drivers. The system is composed of camera modules with microwave sensors and computing units. The type of sensors used in their system are the break-the-beam sensors. The computing unit is composed of a low-cost server, a management system and alert processing, a warning signal on the road, and network equipment (a switch) that makes the interconnection between all of these devices. Further, this architecture contains two cameras, together with two sensors, which capture an animal's image and send it to the management system. This system initiates the machine learning algorithms to detect and classify these images of moving animals. If an animal faces danger due to cars, the particular animal is detected on the road by the management system, which alerts the warning sign on the road for the driver to be cautious. This proposed simulation is the first to integrate the concept of machine learning algorithms in detecting a specific or target

species. Among all supervised algorithms used, KNN shows the best F-measure. However, this work also faces downfalls such as using synthetic images for training the model. The work also experiences less accuracy, with failure in detecting small animals. Moreover, the study is characterized by fewer data with no road tests performed during the testing phase.

It is noticeable that break-the-beam systems effectively detect large animals, which then reduces the collision of large animals and vehicles. The most recent break-the-beam systems are compared and summarized in Table 6. The systems have also registered false positives due to sensors' movement and low visibility. Due to high rates of false positives, drivers attempt to ignore the road's warning signs, mistaking the system to be facing a malfunction. Integration of advanced machine learning algorithms can help make the system intelligent and eliminate false positives, achieving higher accuracy. Equally, adopting the use of state-of-the-art machine learning approaches in RQN4 and integrating them with sensor technology can minimize the false detection and increase the accuracy of the system.

4.3. Buried Cable, Roadkill Detection, Driver Assistance, Autonomous Vehicles, and Mobile Mapping ADSs

Installations of buried cable animal detection systems (BCADS) for wildlife-vehicle detection were in the fall of 2008 along U.S. Highway 160 between Durango and Bayfield [28]. These systems can generate an invisible electromagnetic field that faces disturbance each time a large animal crosses the cable. The disturbance has an adjustable threshold above which it triggers an alarm and activates warning signs, cautioning drivers to be more alert and reduce vehicle speed to help mitigate collisions of cars and ungulates. In 2009, installations of cameras to capture the image of animals crossing the road during both the day and night were successful. The records of images are associated with references such as the date and time stamp. The results of the reliability investigation of the system show $\sim 0.46\%$ of false negatives with no false positives. The percentage of detections of all intrusions in the electromagnetic field (detection area) is relatively high, i.e., ~98.12%. When tested under different environmental conditions, the system generates no false positives and approximately 0.46% false negatives. In 2015, Christian et al. [29] installed a BCADS. The system detects large- and medium-sized animals crossing the road and provides data on their location along the length of the cable. The system generates an invisible electromagnetic detection field around the buried cables. The components of the system are a 300 m long-buried dual-cable sensor and a central processor unit for control and communication which are similar to works by Huijser et al. [28]. In their study, each time the area where detections happen are perturbed, an alarm is declared, and they determine the location of the intrusion. Their work has a surveillance video camera which monitors animal movement and gauges system detections during nighttime and under all-weather conditions. Results show that if properly installed and calibrated, BCADS can detect large and possibly smaller animals with over 95% reliability. However, this system still registers false detections.

Reference	Year	Sensor/Signal Type	Results	Strengths	Weakness
Huijser et al. [17]	2006	Microwave radio	 Detection of an animal when it breaks the beam of light Activated warning signs to alert driver of existence of an animal on the road Increase awareness and reduction in speed of vehicle, hence less collisions 	Increase in the alertness of drivers which leads to fewer or less severe WVCs	Curves, slopes, and vegetation landscapes require additional sensors which increases the cost of the system
Huijser et al. [18]	2009	 Infrared Laser Microwave radio 	Four systems in test-bed are able to detect large animals	 Can detect large animals Excellent visibility, hence low number of false positives 	 Cannot detect small animals False positives due to sensor moving in and out of the alignment Low visibility may block or reduce the narrow signal path of optical break-the-beam systems
Huijser [20]	2010	Microwave	 Detection of animals are on a higher percentage Records date and time of each individual detection Records animal movement within the enclosure 	 Less error rate with 1 false positive and 4 false negatives out of 140 valid detections The system achieves an overall accuracy of 97% 	Challenge in detecting small animals
Grace et al. [21]	2016	• Infrared-beam	 System performs from poor to fair in detecting target animals Flashing light triggers when the beam breaks, this alerts drivers of the presence of an animal on the highway 	• Reduction in the number of false positives by nearly 50% through the use of a control site to test the RADS	 Large number of false positives Encountered system malfunctions that affected the results of the system Ignorance of the system by the drivers, public education, increase in public understanding, and driver awareness is needed to achieve better results
William et al. [19]	2019	Microwave sensors	KNN accuracy greater than rain forest	Method is able to identify animals	 Small sample of dataset System is not tested on real-life images but on synthetic images System was not deployed to conduct more test

Table 6. A comparison of break-the-beam systems in terms of their sensor types, performances, strengths, and weakness.

In 2016, mortality rates of amphibians on Portuguese roads had been rampant [35]. Installation of a roadkill-intelligent system successfully scans for the dead amphibians, hence capturing roadkill records. This system scans the road using a trichromatic line scan camera with light-emitting diode lighting. The two-axle trailer is attached to another vehicle such as a lorry jeep, among others, and pulled. The architecture of the system is composed of an imaging system used for acquiring road surface images: a computer used to control the device, storage, and processing unit that can even operate offline, a GPS for extracting the coordinates for all acquired read frames, a line scan camera and light-emitting diodes used for capturing images, and an encoder used to synchronize all the device. Some of the criteria to follow on implementing such a system are (1) mobility—easy and free system movement, (2) versatility—that is, the flexibility to accommodate all vehicles, (3) motion stability needed for the image capturing system and thus the dual axle system, (4) standalone powering over several hours of utilization, (5) accommodation of all the equipment with protection to rain and dust, and (6) capability to scan the road surface with high resolution of a complete track (\sim 25 km) at an acceptable speed. However, even though such excellent criteria is followed, this system still experiences downfalls such as the lack of machine learning algorithms to automatically detect roadkill of amphibians with high accuracy.

In 2017, Sharma and Shah [31] developed driver assistance to detect animals on highways. Their system uses the histogram of oriented gradients descriptor and cascade classifier to detect videos and images with less computing space. As such, the system achieves a high accuracy due to the ability of cascade classifiers to remove false negatives quickly from the data samples. The algorithm achieves an 80% accuracy when using a small sample of dataset available. Their system does not capture and test the images during the night. Even with these weaknesses, the algorithm accurately detects video and images on the road and alerts the driver whenever the algorithm detects a cow. In the same year, Rosenband [32] discussed the implementation of a self-driving car that requires no human. The car uses software and hardware components to automate navigation. The software helps to simulate and analyze vast amounts of datasets to make self-driving cars a reality. Besides, the hardware contributes to automation through sensing, computing, and embedded control with the help of the software. The sensors collect information such as vehicle surroundings, position, and environment. Later, the sensors send the collected information to a computing unit for processing. The computing unit fuses, processes, and interprets the sensor data. The output of the computing unit is the trajectories. Next, these trajectories pass to embedded control systems, which in turn communicate with the vehicle actuators to manipulate steering, braking, and throttle. The dataset used for detection is images and videos. However, this study is limited with animal datasets, limiting the performance of autonomous vehicles in detecting WVCs. It is worthy to note that less research is conducted on mitigating WVCs using autonomous vehicles.

In 2018, Sillero et al. [33] developed a prototype of a mobile mapping system for classifying amphibians on the road. This system is composed of a scanning system attached to a vehicle. The scanning gadget can record amphibian images on road surfaces up to a speed of 30 km per hour. On evaluating the system, they conduct three tests which show promising results. The first test controls are accompanied by plastic specimens; the second control tests are composed of dead specimens. In addition, the third test is composed of real tests on Mitra campus roads. The data collection process performs analysis and testing on amphibian datasets, achieving acceptable results. The first test correctly classified 94.1%, and failed classification rate is at 5.88%. The second test correctly classified 92.3% dead specimens and failed to classify 7.7%. Finally, with the third test, 83.9% real specimens are classified correctly, and 16.1% failed to be classified. In all cases, the human classification of images achieves 100% correctness. This study is the first to use machine learning algorithms and conduct road tests to detect live and dead amphibians. This work can adapt to any animal group. Unfortunately, the algorithm presented a high rate of false-positive detections.

In 2019, development of a smaller and improved version of the mobile mapping system is successfully implemented in Portugal [34]. It detects dead amphibians and small birds on the road. The new system components include a ZED stereo camera (stereo multispectral and high definition camera), a GPS receptor, a high-power processing computer (GL553VD ASUS laptop), an attachment device, and a lighter charger for vehicles. The manufacturer of the camera component Stereolabs Inc is located in San Francisco, CA, USA. Whereas that of the GL553VD ASUS laptop is ASUS company whose headquarters is at Taipei City, Taiwan. The system runs continuously as long as the vehicle is working. This novel solution is advantageous because of the reduction in the system size, hence being light weight, making it portable and easily attached to the back of any vehicle. Other advantages of this system are more straightforward workflow, unlimited energetic consumption, and sampling width which effectively covers a one-way national road, and the system is considerably cheaper. In measuring the effectiveness of the system, the system detects approximately 78% of the amphibians and birds present on surveyed roads, failing to detect 22% and generating approximately 17% of false positives. This system is applicable and feasible and can improve the implementation of conservation measures. The system saves time for researchers and transportation planning professionals. In the same year, Goswami et al. [110] started developing deep learning algorithms for driver assistance to detect animals and further proposed a dashboard framework to deploy algorithms. The dashboard systems are usually portable and efficiently assist the driver by generating a message signal whenever they detect an animal in the camera frame. Even though this study had a limited dataset, results show that MobileNet-SSD outperforms Mask R-CNN and YOLOv3 in detecting objects.

The Virginia Department of Transportation, in collaboration with the Virginia Tech Transportation Institute, participated in installing BCADS. The system acts as a measure to mitigate the WVCs [30]. The systems are powered by a solar photovoltaic system and a cellular modem provided for remote system monitoring and data collection, installation of a flashing light—"deer crossing"—warning sign at the site, and wirelessly linked to BCADS to alert approaching drivers whenever a detection of an animal is observed. These systems provide roadside or in-vehicle warnings if an animal is active on or near the road. The system's performance is monitored on a public road to assess real-world system capabilities and operational issues. On evaluation under controlled and secure conditions, there was successful reliability and effectiveness of BCADS. These systems identify crossings of largeand medium-sized animals with approximately 99% reliability and provide data on their exact location along the length of the sensing cable. Under all-weather conditions, BCADS collected data to monitor animal movement, vehicle traffic, and system performance. The authors collect data on driver response; warning signs are activated during the dawn and dusk hours, and the data shows that approximately 80% of drivers either braked or slowed in response, indicating that the signs are effective. However, we also see that 20% of the drivers are not complying with the warning signs.

In summary, although existing intelligent ADSs detect large animals, these systems still register a lot of false detections. Generally, the systems fail to distinguish between the movements of large animals and other objects. Besides, they fail to restrict detections to only wildlife movement. With this, these systems require algorithms that will help them distinguish between different objects or help recognize target objects, hence reducing the errors that lead to false positives and false negatives. Moreover, todate, there is no work that uses geophones, UAVs, GPS, and VHT tags to detect animals with a motive of mitigating WVCs. Table 7 summarizes the buried cable, roadkill detection, driver assistance, autonomous vehicles, and mobile mapping ADS. In Section 5, we discuss the different datasets used to prevent WVCs.

Reference	Year	Application	Results	Strengths	Weakness
Huijser et al. [28]	2012	BCADS	• All intrusions in the detection area are detected with a high percentage of 98.12%	 The system responds quite well when tested under different environment conditions, it generated no false positives and very few false negatives False positives are 0% on testing reliability of the system 	• Reliability of the systems results are still not favorable on the number of false negatives (0.46%)
Druta et al. [29]	2015	BCADS	 System detects large- and medium-sized animals crossing the road Provides data on the location of animal detected along the length of the cable 	• Detects animals with 95% accuracy during nighttime and under all-weather conditions	 System still registers false detections In some cases, when a vehicle crosses the edge line and was entering the detection field, an alarm triggers and is declared as a regular intruder detection. To some extent, the signal response depends on the vehicle distance from the detection field Nuisance alarm triggers are due to water movement on the cable path
Gil et al. [35]	2016	Road-kill detection	 Scan images of amphibians on-road and GPS coordinates Capture amphibians and road surface images Data storage and data processing All these devices are synchronized using an encoder attached to the left rear wheel of the trailer 	 Mobility Versatility in being easy to attach to any vehicle Motion stability needed for the image capturing system and thus the dual axle system Standalone powering over several hours of utilization Accommodation of all the equipment with protection to rain and dust Capability to scan the road surface with high resolution of a complete track (~25 km) at an acceptable speed 	• Need to increase the accuracy of the system, on using the Haar classifier method of computer vision which achieves 65% accuracy which was low
Sharma and Shah [31]	2017	Driver assistance	 The algorithm can detect an image, video, and the distance between the animal and car. Send alerts to the driver distance away before the animal and car move closer to each other, hence minimizing collisions 	• The algorithm can detect animals in both a video and image	 The study uses limited dataset, hence need for improved performance Not tested object detection during the night
Rosenband [32]	2017	Autonomous vehicle	Delivers autonomous self-driving without requiring humans	• These cars are able to detect objects on the road by adapting to different environment	 The cars have registered accidents with humans, animals, etc., and also maneuvers [111]

Table 7. Summary of results, strengths, and weaknesses of BCADS, roadkill detection, driver assistance, autonomous vehicles, and mobile mapping ADSs.

Table 7	. Cont.
---------	---------

Reference	Year	Application	Results	Strengths	Weakness
Sillero et al. [33]	2018	A road mobile mapping device for supervised classification	• The algorithm is able to detect amphibians from the amphibian dataset in all the three cases	 The classification algorithm can adopt to any classification group The system can work day and night It is ideal for passive surveys 	The work still faces false-positive detection
Guedes et al. [34]	2019	Mobile mapping system 2	 System detects approximately 78% amphibians and birds (overlooking 22%) 	 The survey can be conducted by any person using mobile mapping system 2 with or without sampling skills 	Generated 17% of false positives
Goswami, et al. [110]	2019	Driver assistance	 Three algorithms (Mask R-CNN, YOLOv3, and MobileNet-SSD) are able to detect animals on the road 	MobileNet-SSD performed better in detecting animals	 The study uses a small dataset The algorithm is not deployed on the driver assistance and tested
Druta et al. [30]	2020	BCADS	• The system identifies crossings of large- and medium-sized animals and provides data on their location along the length of the sensing cable	 BCADS prove to be effective and reliable when evaluating under controlled and secure conditions Sometimes the system is able to detect small animals Able to detect small, medium-sized, and large animals with approximately 99% reliability 	 Total 20% of the drivers do not respond to the warning signs, which is dangerous, as this can lead to WVCs False negatives—this is when a target crosses over the cable or is inside the detection zone and an alarm is not triggered

5. RQN3: What Types of Datasets Are Currently Used to Mitigate WVC?

In this section, we provide a survey of the available datasets used in the primary literature to mitigate WVCs. Moreover, the section suggests datasets to use in mitigating WVCs. For each of the datasets, we identify different tasks implemented in the primary studies to mitigate WVCs, and we propose additional tasks to apply to the dataset to further minimize WVCs. Data collectors such as citizen scientists, experts, and Kaggle collect the identified datasets. Table 8 shows parameters captured and these are name of dataset, type of data, task executed or proposed, collectors, data collection period, the number of instances, country, and the reference works. The datasets in Table 8 are presented in alphabetical order and are composed of time series, image, and sound datasets. The label \checkmark indicates that the work at the "Ref works" column has implemented the specified task. The symbol \times means a specific task is proposed to be implemented with the dataset, but we did not find any work that implements the task in the literature. The label – indicates that the authors are not explicit about a specified parameter value in the column. We note that we did not find reference works for rows labeled as "None".

The time series datasets are major records of WVC instances. For time series, the collectors capture the name and the number of species killed in a specific time frame. Datasets such as WVC hotspots [112] and snake roadkill [39] predict WVCs. Possibilities of other machine learning tasks that may use these two datasets are classification and forecasting. The WVC dataset [112] contains a count of 529 species killed in Italy. The data are collected from 2014 to 2016 by citizen scientists. Predictions of WVCs are observed by Valerio et al. [112] while using this dataset. In Taiwan, citizen scientists collect and record \sim 40,000 instances of snake roadkill [39]. Later, the dataset helped Yue et al. [39] to predict the death of snakes on the road. The collection period is from 2006 to 2017. It is worthy to note that time series datasets such as animal-related crash data [113] and roadkill datasets [114–116] all focus only on collection but never perform any machine learning techniques on the data. Animal-related crash data [113] are collected by experts from 2001 to 2007. Every year, 0 to 4285 instances are collected in Australia. From 2007 to 2008 in Kenya, data on roadkill [114] are collected. The collectors record 1436 instances for 164 days over an 11 year period. We find no experiments performed on this dataset. Therefore, implementation of machine learning models is vital to perform experiments. Besides, in Tanzania, roadkill data [115] collections happened between 1990 and 1991. The experts collected 183 instances. In South Africa, the roadkill dataset collected from 2011 to 2014 helps mitigate WVCs [116].

Image datasets such as Pothole dataset [117], Stanford cars dataset [118], Oregon wildlife dataset [119], and Kangaroo dataset (https://www.kaggle.com/hugozanini1/kangaroodataset, accessed on 14 August 2021) are vital in mitigating WVCs. These majorly capture examples of the different animal images used in a machine learning model or framework. They perform object detection tasks with other tasks such as classification, segmentation, and localization not exhaustively applied. Fan et al. [120] use the Pothole datasets [117], Oregon wildlife dataset [119], and Stanford cars dataset [118] in their study to mitigate roadkill. The latter dataset detects and distinguishes vehicles from other objects. These three datasets are collected on different occasions and in different locations for the use of object detection. In 2018, in the USA, a dataset composed of a collection of cars was collected by Kaggle and referred to as Stanford cars dataset [118]. It is a large dataset with approximately 16,185 records of car images. The Oregon wildlife dataset [119] is collected from Oregon in 2019. The dataset is open source and published on Kaggle for easy access. It contains 2029 image instances. In the same year, 2019, Pothole image dataset [117] is collected by scraping online images.

Additionally, object detection is further being applied to Kangaroo dataset (https://www.kaggle.com/hugozanini1/kangaroodataset, accessed on 14 August 2021) and Serengeti dataset (https://lila.science/datasets/snapshot-serengeti, accessed on 20 September 2021). In 2020, 313 images of Kangaroo are scraped online to collect a Kangaroo

dataset (https://www.kaggle.com/hugozanini1/kangaroodataset, accessed on 14 August 2021). Later, the images are annotated to build machine learning pipelines for object detection. Object detection achieves tremendous results on this dataset when applied by Yi [41]. In addition, other machine learning tasks such as classification, segmentation, and localization may apply to this dataset. Another fundamental dataset is the Serengeti dataset (https://lila.science/datasets/snapshot-serengeti, accessed on 20 September 2021) which contains 7.1 million images collected by conservationists in 2010 in Tanzania. It contains a variety of animal species used to mitigate WVCs when modeled with machine learning. Some works by Marco et al. [121] and Sadegh et al. [122] design object detection algorithms that detect and count animals in wild reserves. The image dataset YOLOv3 achieves high accuracy on an image localization task by Redmon et al. [123]. Moreover, the dataset contains images of animals used to mitigate WVCs. In 2020, 1000 instances of Cheetah, Hyena, Jaguar and Tiger dataset (https://www.kaggle.com/c/swdl2020/data, accessed on 1 September 2021) are collected by experts. This dataset has a good sample size, prompting us to recommend tasks such as object detection, localization, classification, and segmentation.

Other image datasets such as YOLOv3 (https://pjreddie.com/darknet/yolo/, accessed on 9 December 2021) and Waymo open dataset (http://www.waymo.com/open, accessed on 11 December 2021) have performed with acceptable results in localization, and detection and tracking tasks, respectively. YOLOv3 (https://pjreddie.com/darknet/yolo/, accessed on 9 December 2021) dataset used localization techniques [123] to mitigate WVCs. Waymo open dataset [124] (http://www.waymo.com/open, accessed on 11 December 2021) is a state-of-the-art image latest image dataset for autonomous driving. The dataset contains 1150 scenes, each of which spans 20 s. It is composed of images of calibrated high-quality LiDAR and camera data. The tasks that use this dataset are detection and tracking.

Sound dataset such as wildlife–train collisions data [125] are vital in mitigating WVCs. This particular dataset is collected from different sensors whenever a train approaches a hotspot area. The train uses an approach detector which uses vibrations in the rail to detect trains and creates a warning system. When the warning activates, animals leave the railway, hence mitigating roadkill cases.

We find that some state-of-the-art datasets, such as those used in autonomous cars, do not capture animals as a sample in the training examples, e.g., Kitti dataset [126]. Because of this, there is a need for a balanced, fair dataset for detecting animals. This balanced, unbiased dataset increases the accuracy of detecting animals in autonomous vehicles' ADSs. Other systems such as break-the-beam, area-cover, and driver assistance can embrace such image datasets highlighted in Table 8.

Table 8. Summary of datasets that are used in mitigating WVCs showing types of dataset, task (\checkmark means task is performed on data and \times means a task we have proposed that can be performed on data and no work in the literature has performed the task), collectors, collection period, number of instances, and country (– means authors did not state data collection period, instances, and country).

Dataset	Type of Data	Task	Collectors	Collection Period	#Instances	Country	Ref Works
Animal-related crash data [113]	Time series	 Prediction × Forecasting × Classification × 	Experts	2001 to 2007	0 to 4285 per year	Australia	None
Cheetah, Hyena, Jaguar and Tiger dataset (https: //www.kaggle.com/c/swdl2020/data, accessed on 1 September 2021)	Images	 Object detection × Classification × Segmentation × Localization × 	Experts	2020	1000	-	None
Data on roadkill [114]	Time series	 Prediction × Forecasting × Classification × 	Experts	2007 to 2018	1436	Kenya	None
Kangaroo dataset (https://www.kaggle.com/ hugozanini1/kangaroodataset, accessed on 14 August 2021)	Images	 Object detection ✓ Classification × Segmentation × Localization × 	Kaggle	2020	313	Scraping online	Yi [41]
Oregon wildlife dataset [119]	Images	 Object detection ✓ Classification × Segmentation × Localization × 	Kaggle	2019	2029	Oregon	Fan et al. [120]
Pothole image dataset [117]	Images	 Object detection ✓ Classification × Segmentation × Localization × 	Kaggle	2019	618	Scraping online	Fan et al. [120]
Roadkill data [115]	Time series	 Prediction × Forecasting × Classification × 	Experts	1990 to 1991	183	Tanzania	None
Roadkill dataset [116]	Time series	 Prediction × Forecasting × Classification × 	Citizen science	2011 to 2014	2642	South Africa	None
Serengeti dataset (https: //lila.science/datasets/snapshot-serengeti, accessed on 20 September 2021)	Images	 Object detection ✓ Classification × Segmentation × Localization × 	Experts	2010	7.1M	Tanzania	Marco et al. [121], Sadegh et al. [122]

Table 8. Cont.

Dataset	Type of Data	Task	Collectors	Collection Period	#Instances	Country	Ref Works
Snake roadkill [39]	Time series	 Prediction √ Forecasting × Classification × 	Citizen science	2006 to 2017	>40,000	Taiwan	Yue et al. [39]
Stanford cars dataset [118]	Images	 Object detection ✓ Classification × Segmentation × Localization × 	Kaggle	2018	16185	USA	Fan et al. [120]
Waymo open dataset [124] (http://www.waymo.com/open, accessed on 11 December 2021)	Images (LiDAR and camera)	• Detection and tracking \checkmark	Experts	_	1150	USA	Sun et al. [124]
Wildlife-train collisions data [125]	Sounds	• Train detection \checkmark	Experts	_	183	Canada	Backs et al. [125]
WVC hotspots [112]	Time series	 Prediction √ Forecasting × Classification × 	Citizen science	2014 to 2016	529	Italy	Valerio et al. [112]
YOLOv3 dataset (https://pjreddie.com/darknet/yolo/, accessed on 09 December 2021)	Images	 Object detection × Classification × Segmentation × Localization √ 	Experts	2018	_	USA	Redmon et al. [123]

6. RQN4: What Types of Machine Learning Algorithms Are Used to Mitigate WVCs?

In 2013, Parikh et al. [127] used MATLAB to implement template matching and motion detection using the frame difference method to detect birds on the road. While detecting the motion of objects in the video, denoising for the frame and several frames is initiated on a two-frame difference. Additionally, researchers perform background subtraction to obtain a binary image [128] and subtract two frames to attain differencing. These subtractions are mostly performed pixel wise. This system operates by using stored video and background subtraction methods. Features extraction is from the stored video and database. Later, the features generate target and template images which are fed into a template matching algorithm for detection. Template matching is a technique in digital image processing for finding small parts of an image that match a template image. While implementing template matching, authors adopt the concept of normalized cross-correlation. On measuring the similarity of the images, the results varied from -1 to 1 with 1 meaning that the image is identical, -1 that the image is the negative of the other one, and 0 that the two immediate images do not correlate. When the bird is detected, a generated alarm triggers as feedback. On testing, the background subtraction methods perform well in detecting birds. The false-positive and false-negative for bird detection are 5.94 and 5.23%, respectively. This technique had 94% efficiency in detecting birds.

In 2017, Tibor et al. [38] created an animal dataset that consists of 500 different subjects (5 classes with 100 images for each class). The five classes are bear, hog, deer, fox, and wolf. Recognition and classification of the different animal species using CNN are successful. Other algorithms compared with CNN in recognition of animals are those such as: (1) principal component analysis (PCA), in which the PCA reduces the dimensionality of large datasets by transforming a big set of input variables into a smaller one with almost all the information obtained in the big set present [129]; (2) linear discriminant analysis (LDA), which reduces the dimensionality of the data by maximizing the separability among known patterns [130]; (3) local binary patterns histograms (LBPH), an image recognition algorithm that uses pixel's 3×3 neighborhood to extract the LBP code and in which the process involves thresholding the 3×3 neighborhood of each pixel with the center value using mean, median, and center pixel as thresholds [131]; and (4) SVM which classifies the different species. Results show that the CNN model achieves the best accuracy of 97% on a bear, 95% on a wolf, 95% on a fox, 93% on a deer, and 91% on a hog. The highest accuracy scored by other models is 82% on bear and hog by PCA, 83% on a hog by LDA, 87% on wolf by LBPH, and 87% on a bear by SVM as summarized in Table 9. Even though this study did not test this application on the roads, we highlight that the work is vital in recognition and could classify animals on roads when integrated with an alert system to minimize WVC.

Accuracy of Correctly Identified Animals for Each Class (%)					
	Bear	Wolf	Fox	Deer	Hog
РСА	82	79	78	76	82
LDA	81	77	78	81	83
LBPH	85	87	83	84	82
SVM	87	864	85	83	81
Proposed CNN	97	95	95	93	91

Table 9. Compared results of correctly classified animals from an animal dataset with 5 classes (Results from Tibor et al. [38]).

Antonio et al. [19] developed a methodology that detects animals using animal images captured by roadside installed cameras. This methodology allows features extraction of regions of the image and further uses machine learning techniques to classify areas into two classes: animal and non-animal. In this work, the KNN and random forest (RF) algorithms achieve acceptable generalization. KNN ranks a test sample according to the class of its *K* nearest training samples and also defined according to a distance metric. RF generates randomness for a model when the trees grow. Instead of finding the best feature of splitting a node, it looks best among a random subset of resources. Results show that KNN had a slightly higher accuracy of 0.6243 when compared to RF with an accuracy of 0.6219. However, adoption of more robust machine learning models is vital to help achieve accuracy greater than 0.6243.

In 2019, mobile mapping system 1 was upgraded to mobile mapping system 2 in Portugal [34]. Mobile mapping system 2 includes state-of-the-art deep learning image classifiers such as (1) VGG16, a CNN model with a 16 weight deep layers network of a stack of convolutional layers, fully connected layers, softmax layers, and rectified linear activation function [132], (2) VGG19, which contains 19 weight network layers of 16 convolutions and 3 fully connected layers, in which an image passes a stack of convolutional layers, max-pooling layers, fully connected layers, softmax layers, and rectified linear activation function to be recognized [132], (3) ResNet50, a 50 layer residual network stacked and trained for a particular task, it learns many features at the end of its layers using skip connections, and the residual is a subtraction of a feature learned from an input of a layer [133], (4) Inception-V3 which modifies the previous inception models by using less computational power [134], and (5) Xception, a deep CNN architecture that depends on depthwise separable convolutions layers. Depthwise separable convolutions are composed of depthwise and pointwise convolution [135]. These image classifiers tested on dead amphibians and birds have achieved tremendous results. Real and controlled surveys help to evaluate the system's performance. Controlled surveys use a dataset of the dead specimen from the University of Evora, and researchers conduct real surveys on the national roads of Evora in real-time. In real surveys, detecting dead species is approximately 78% of the amphibians and birds on Evora roads, with a false negative of 22% and roughly 17% of false positives. Controlled surveys detections are between 63.3 and 80% of animals. The false positives are between 17.4 and 24% and false negatives between 20 and 36.7%. In East Asia, mortality rates of different species of snakes dataset are citizen-collected from Taiwan [39] as shown in Table 8. The maximum entropy environmental niche modeling helps in predicting the snake roadkill. This method finds the probability distribution that best fits data [136]. Results show that patterns across snake species differed in habitat use, foraging behavior, and taxonomic group. Roadkill sightings are highest in forests and strips of farmland or shrubland that cut through forests; these areas likely support high snake abundances or dispersal activity. Sightings are lowest in urban areas, likely because of unfavorable habitat conditions. Road density has little influence on roadkill sightings. Dense roads may be associated with lower habitat quality, hence containing fewer snakes. To achieve high accuracy in mitigating snake roadkill efficiently, considerations of natural history and landscape factors are vital.

In 2020, a dataset composed of various image samples is split into 75% train and 25% test and trained on a CNN vision algorithm [40]. The vision algorithm detects and classifies a cheetah at 79% accuracy and an elephant at 86% accuracy. It uses a four-layer approach to train the input images, e.g., convolutional layer that extracts features from an input image, rectified activation function, pooling layer, and fully connected layer. In the same year, Yi et al. [41] innovated and developed a roadkill alert system called ROOD. This system alerts drivers of high-risk areas before they encounter animals. It does this by detecting dead animals on the road and sending signals to drivers by using LED lights and vibration to detect the roadkill. If the driver detects the roadkill as a true positive, the driver sends the GPS location and photos back to the database. The reported GPS location sets up an instant alert to the nearby drivers, and the location is marked a high-risk area based on the collision rate. The dataset used in this study is YOLOv3 (https://pjreddie.com/darknet/yolo/, accessed on 9 December 2021) and Kangaroo dataset (https://www.kaggle.com/hugozanini1/kangaroodataset, accessed on 14 August 2021) for training the YOLOv3 algorithm [123]. The algorithm uses residual block, bounding box regression, and intersection over union techniques. The image is divided into

many grids of equal dimensions for the residual blocks. Every grid cell will detect objects that appear within them. For example, if an object center appears within a specific grid cell, this cell will be responsible for detecting it. The bounding box outlines and highlights an object in an image. The bounding box attributes are width, height, class (for example, lion, elephant, etc.), and bounding box center. YOLOv3 uses a single bounding box regression to predict objects' height, width, center, and class. Later, the algorithm represents the probability of an object appearing in the bounding box. Intersection over union is a metric in localization that describes how boxes overlap. Each grid cell is responsible for predicting the bounding boxes and their confidence scores. High performance is when the intersection over union is equal or near to 1, and poor performance is when the intersection over union is equal to 0 or closer to 0. The study allows road authorities and biologists to respond to the collisions immediately and assists in planning road infrastructure improvements. This system was not installed on real cars so that each vehicle turns into a peri-urban citizen. Adequately, this research suggests three potential directions for mitigating roadkill: changing animals' behavior, improving road infrastructures, and changing driver behavior.

In summary, this section discusses works that use machine learning algorithms to mitigate WVCs as shown in Table 10. Figure 6 captures the intelligent systems and the different tools and devices integrated into these intelligent systems to mitigate WVCs. We also note in Figure 6 the various types of machine learning algorithms currently implemented to detect animals and to mitigate WVCs. Moreover, we find that most machine learning algorithms used to detect animals in wildlife do not mitigate WVCs but monitor and count species. Besides, there is non-exhaustive research on mitigating WVCs using reinforcement learning agents. As well, no identified research is exhaustive on the integration of machine learning with ADSs such as area cover and BCADS. In Section 7, we discuss issues arising from the review and suggestions on filling in the gap.



Figure 6. Summary of existing intelligent systems and/or machine learning methods that prevent WVCs.

Reference	Year	Method	Data	Application	Results	Strengths	Weakness
Parikh et al. [127]	2013	 Feature extraction Frame differencing Template matching algorithm 	 Videos of birds Target images Template images 	• Real-time application	 A 94% accuracy in detecting birds A 5.94 % false positive rate for bird detection A 5.23% false negative rate for bird detection 	 Bird detection with high accuracy This real-time system can be adapted to detect large animals before they enter the road to help mitigate WVCs May help to save crops in farm from animals, hence mitigating human–wildlife conflicts 	System still experiences false detections
Tibor et al. [38]	2017	 CNN PCA LBPH SVM LDA 	• Images of wolf, bear, deer, fox, and hog	• Animal recognition system	 Highest accuracy scored models: 97% on bear by CNN, 82% on bear and hog by PCA, 83% on hog by LDA, 87% on wolf by LBPH, and 87% on bear by SVM 	Highly predicted the class labels	No domain applicability statedFailed to classify some animals
Antonio et al. [19]	2019	KNNRandom forest	Synthetic images	Simple ADS	• KNN accuracy is greater than random forest	• Method is able to identify animals on the road	 Dataset was too small Images used are not real-life The system is not deployed and tested on roads The accuracy of the system was slightly above average
Guedes et al. [34]	2019	 CNN VGG16 VGG19 ResNet50 Inception Xception 	• Images of amphibians and birds	• Mobile mapping system 2	 Detects 78% amphibians Overlooked 22% birds A total 17% false positives The algorithm classified 37.1% of the detected animals to the species level (the remainder was classified as "others") 	 Mobile mapping system 2 better than previous system mobile mapping system 1 High detectability Detects small animals 	 It still requires human intervention to determine if the animal is alive or road-killed The weather conditions may affect the images' quality The system is not suitable to use at night: there is no light system as in the previous version A high storage capacity is necessary (the tests originated 1.65 terabytes of images) Specialized people needed for image processing and animal identification A workstation is required for computing heavy data

Table 10. Comparison of machine learning methods used to mitigate WVCs.

Table	10.	Cont.
-------	-----	-------

Reference	Year	Method	Data	Application	Results	Strengths	Weakness
Yue et al. [39]	2019	Maximum entropy environmental niche modeling	Roadkill snakes images	• Snake roadkill mitigation	Detected road-killed snakes	 They sighted patterns across snake species differing in habitat use, foraging behavior, and taxonomic group 	 Natural history and landscape factors contribute a lot to the increase in number of snake roadkill
Banupriya et al. [40]	2020	• CNN	• Wildlife images, e.g., elephants and cheetahs	• Animal detection	• The model detects a cheetah at 79% accuracy and an elephant at 86%	• Able to classify animals correctly with a good accuracy	No warning to the driver about the detected animalAlgorithm was not tested on roads
Yi [41]	2020	• YOLOv3	 Kangaroo dataset (https://www. kaggle.com/hugozanini1 /kangaroodataset, accessed on 14 August 2021) YOLOv3 	• ROOD System	 Developed a roadkill alert system that mitigates roadkill and alert drivers of high risk areas 	• Data collected help policy makers, road authority, and biologists to respond to collisions	• This system is not installed on real cars

7. RQN5: What Are the Limitations of the Reviewed Studies in Mitigating WVCs? What Are the Proposed Solutions?

Many attempts highlighted in Sections 4 and 6 minimize WVCs. To date, roadkill continues to be rampant due to prevailing conditions in Section 3 that contribute to the occurrence of WVCs. Moreover, implemented solutions discussed in Sections 4 and 6 are not effective in reducing roadkill due to high rates of false detections. We highlight, in-depth, the issues and challenges arising from the review and the way forward as summarized in Table 11.

Current Issues and Challenges from the Review	 Current systems are not suitable for small and medium-sized animals Machine learning algorithms do not detect individual species and their behaviors but generalizes species High false detection of existing systems Some systems fail to alert drivers about the presence of animals on the road Failure to detect hotspot areas Investigations on ethics in AI in wildlife conservation is not explored Current ADSs are expensive to implement ADSs do not exist in South Africa and Africa Limited data sets for training AI models MMS needs human intervention to detect small dead animals Landscapes with curves, e.t.c. need more hardware tools Driver ignorance on the usage of roads and ADSs
Proposed Future Research Directions	 Design matching algorithms that detect species vulnerable in specific season and month of the year An update and upgrade of the existing system is essential Improved vision algorithms to detect each species Development of other new and improved efficient solutions Incorporating connectivity in wildlife reserves with ADSs as a consideration Sensitizing drivers about ADSs Enable open data for research in wildlife conservation Develop safe, reliable, and robust ADSs that is fair and accountable Venture into other machine learning methods and ADSs, e.g., geophones, UAVs, GPS, and VHT tags. Integrate them to work towards mitigating WVC Development of real-time ADS Conduct exhaustive primary studies on factors and sub-factors contributing to WVC Use 3D point cloud data sets in mitigating WVC Detecting WVC hotspots

7.1. Current Issues and Challenges Arising from the Review

Although these systems record tremendous results in mitigating WVCs, they still experience enormous issues and challenges:

- Most current ADSs are not suitable for small- and medium-sized animals. We note that small animals are killed more by WVCs than big animals, which requires urgent attention in implementing tools that mitigate WVCs for all animal species. Most solutions concentrate on creating solutions that detect large animals, leaving the non-large animals vulnerable to roadkill.
- The machine learning algorithms do not understand each animal species and its behaviors but generalize species. Developing machine learning models that understand and detect animal behavior of each species is vital.
- There is high false detection due to bad weather conditions and thick vegetation cover, among other factors. The detections are either false positive detections (where the system reports a detection and there is no animal) or false negative detections (where the system did not report any detection yet there was a large animal present) or both.
- Some current systems fail to create an alarm to alert the driver whenever the system detects an animal. In addition, other systems experience delays in detection and sending alerts.
- The embedded solutions implemented so far are usually expensive. There is no low-cost hardware that mitigates WVCs.
- There is a gap in the investigations of fairness in Artificial Intelligence (AI) [137], without excluding AI for the wildlife conservation sector [138]. Auditing and explana-

tion of machine learning algorithms is vital to ensure algorithms observe fairness in machine learning.

- ADSs do not exist in South Africa and Africa and mostly exist in the Northern Hemisphere.
- Limited datasets for training machine learning models to detect animals on the road.
- Mobile mapping systems need human intervention to detect dead amphibians and birds. Moreover, the system captures poor quality images during bad weather seasons.
- The ignorance of drivers and other road users on the usage of the existing ADSs greatly limits systems to achieve acceptable results during system testing.

7.2. Future Research Directions and Way Forward

Our review being investigative and interpretive raises opportunities for future research directions. It does this by identifying challenges in Section 7.1 and then proposes future research directions. Implementing the proposed solutions as stated in Section 7.1 depends on the demands of new learning environments. In this section, we discuss the future research directions as listed below;

- We suggest the design and development of a matching algorithm(s). The algorithm(s) should match species to season(s) or month(s) of the year when vulnerable identified species make frequent crossing attempts toward the opposite side of the road. It is worthy to note that different animal species possess different behaviors during different seasons. Because of these fluctuations in animal behavior, each animal species will frequent the roads during different times and seasons. Animal movement dataset is vital when designing the matching algorithm. This is because the dataset records location coordinates, animal name, sensor type, and timestamp [139]. Further, drivers can be sent alerts each time the algorithm matches a specific species to season(s) or month(s). The notifications should tell the driver that there is a high probability that a specific species is on the road or near the road. The alerts can be sent on the mobile client or dashboard through GPS. Besides, these machine learning algorithms can use factors such as climate to alert the expected presence of animals toward roads during a specific period of the year. Proposal of using a movement dataset is possible because this kind of dataset, together with machine learning models such as hidden Markov models, has shown promising results in classifying animal behavior and estimating the location of species by GPS telemetry [140].
- We recommend upgrading or updating existing systems. The existing systems may be modified and integrated with cutting-edge improved vision algorithms to detect each animal species and classify it with classical animal monitoring system capabilities. Such a solution will increase the accuracy of the already implemented systems in preventing WVCs. Most of these current systems implemented are either break-thebeam, area cover, or buried cable ADSs using sensor technology. However, combining these sensor technologies with vision algorithms is vital and helps detect specific individual species, hence minimizing the false detections.
- We propose development of a real-time roadside ADS as shown in Figure 7 in Africa. This system should monitor and track animals on the road. It is composed of (1) a power supply, supplying the whole system with power. It supplies the computing unit, camera, and sensors, (2) a switch to connect all the hardware devices or system, and (3) photoelectric sensors having an emitter and receiver. The emitter sends a beam of light to the receiver. When an animal crosses the light beam, the receiver alerts the emitter about the presence of an animal. It also includes a (4) camera—when the emitter sensor is alerted of a presence of an animal on the road, it invokes the camera to take the picture. On taking the picture, the camera saves this information to the computing unit (5) raspberry Pi, which is configured together with machine learning algorithms to detect animals on the roads. The computing unit should be configured with protocol buffers or Python to enable system interconnection. As well as communication between the hardware and software and (6) warning signal device

(LED-matrix display), the device is placed on the road for the drivers to read and be notified if an animal is crossing the road. When the picture of an animal is taken by the camera, the computing unit processes the images using machine learning algorithms. The LED-matrix display sends an alert on the highway if it detects the image. The system can further be implemented as a reinforcement learning agent to operate like the real-time roadside animal detection proposed system.

- We suggest developing other new, improved, and efficient solutions. The solutions should be different from the real-time roadside ADS proposed above; such solutions can be, e.g., (1) a system that extends to send an alert to police in case of a collision to clear up the road. The notification can be a text message or a mobile application that pins the location of the accident remains. The driver or stakeholders in the community can send the location of the accident too. (2) Driver assistants can be modified to use AI in wildlife conservation where an agent can detect a hotspot based on existing factors such as forests underpasses. In addition, the agent may predict the collision of animals and cars in a particular region based on factors such as weather patterns, among others. (3) Improvement of mobile mapping system 2 and the development of new solutions that focus on detecting small animals will minimize WVCs.
- Incorporating connectivity in wildlife reserves with ADSs. Connectivity in wildlife is composed of habitat connectivity which looks at the degree of movement of organisms or ecology processes, i.e., the more the movement, the more connectivity, and the less movement, the less connectivity. Landscape connectivity looks at how the landscape facilitates species movement and other ecological flows. It is important to deploy the ADS (moveable structure) at a location with high connectivity. These systems will improve detecting animals since the exact location of system deployment experiences high animal movement patterns. Such considerations will drastically contribute to eliminating roadkill. We note that a swift change in the location of moveable structures to new locations is possible should connectivity change due to different animal behaviors and change in habitats.
- Sensitizing drivers about ADSs. Training drivers about road usage and also the
 presence and usage of the ADS in hotspot areas is vital. This will result in reducing
 unnecessary WVCs, which are caused majorly by the ignorance of drivers about the
 ADSs on the roads.
- Open data: We recommend that conservation agencies collecting data from these existing wildlife monitoring tools loosen their rules for accessing this data for research purposes. Additionally, emphasis must focus on having a fair and balanced open dataset in wildlife conservation to ease the development of cutting-edge research in mitigating WVCs.
- Developing safe, reliable, and robust ADSs that use machine learning. The designs must be fair by ensuring bias is minimized through algorithm audits [141] and other AI ethics practical implementations. Moreover, the fairly designed algorithms should reduce privacy and security attacks, e.g., model extraction attack, model inversion attack, poisoning attack, and adversarial attack on an AI system [142]. Globally, researchers emphasize technology to be made more human-centric. For AI to be fair, work on explainability of AI [143,144] is on the rise to minimize challenges raised by centralized machine learning architectures such as privacy concerns and failure to explain and interpret the AI models. Additionally, scrutiny of AI algorithms is escalating due to the mistrust of AI, leading to the high demand for auditing frameworks [141]. Attempts made help minimize road accidents by applying AI fairness to the health sector [145]. Moreover, documented guidelines on integrating AI ethics in wildlife conservation AI systems help researchers consider AI ethics in wildlife wVCs leading to safe, reliable, and robust ADSs.
- Venture into developing or upgrading existing ADSs and machine learning methods used in animal detection. Combine these to detect animals and mitigate WVCs. The

ADSs that detect animals are geophones, UAVs, GPS, and VHT tags. Integrating these with deep CNN and other machine learning methods used so far in the detection of wildlife is vital and helps mitigate WVCs.

- Although progress is made in understanding the factors contributing to WVCs, there
 is still a paucity of data and information associated with not only factors but also subfactors leading to the occurrence of WVCs. We propose future research to exhaust both
 positive and negative factors and sub-factors contributing to WVCs, hence generating
 enough literature for the secondary studies. Factors reviewed are climate change and
 season, animal behavior, human behavior, and road features. Proposed sub-factors
 under road features are road-type, width, and curve. For animal behaviors, we propose
 innate behaviors and learned behaviors. Sub-factors for climate change, season, and
 human behaviors are vital when exploring how these factors and sub-factors lead to
 WVCs. The factors that lead to WVCs have commonly been non-exhaustively studied
 in isolation. However, a more integrated approach in primary studies is needed to
 examine the factors leading to WVCs.
- State-of-the-art point cloud datasets together with machine learning are achieving acceptable results and are good in detecting, classifying, and segmenting objects on the road. However, these datasets are not used comprehensively in mitigating WVCs. This gap needs to merge since a huge data collection of point clouds data samples by the LiDAR sensors act as the eye of self-driving vehicles providing a 360-degree view of their surroundings to enable safe driving and hence less WVCs.
- Detecting WVC hotspots helps to inform planning to improve driver and wildlife safety. WVC hotspots can be detected using machine learning. Whenever a driver or any other stakeholder is involved in an accident or is in contact with an accident, they can take a picture and save the details on a server using a mobile application. The system's interface should capture the image and location of the accident. Moreover, WVC hotspots detection may be by collecting road features image and video data, training computer-vision machine learning models, and deploying such models on the car dashboard. Whenever a driver is in the car and arrives at the hotspot area, the system should be able to alert the driver to reduce the speed.



Figure 7. A proposed real-time roadside ADS with a computing unit that has machine learning models to detect animal species captured by the camera unit.

8. RQN6: How Applicable and Feasible Are the Proposed Solutions?

The proposed solutions are applicable because they come as a solution to minimize the issues and challenges arising from the review, hence mitigating WVCs. These solutions are applicable because of the following advantages they come with as illustrated in Figure 8:

 Road damage: On implementing the proposed solutions, we attain good results such as (1) minimizing road damages caused by rampant WVCs between animals and cars in hotspot areas, (2) minimizing costs that are associated with the road users such as transport fares charged to passengers as well as the rate at which the car components need replacement, (3) improvement on the market for farmer's products by limiting road damage, which increases mobility and eases farmers in accessing markets for their products, (4) eases the transportation of imports and exports which later improve the country's gross domestic product, and (5) the citizens will start experiencing less road traffic in hotspot areas.

- Accident car collision: The proposed solutions minimize not only the escalating number of WVCs but also the damage they impart on properties. Properties such as cars, items within the cars, and properties usually near the collision sites such as bridges, signposts, electricity poles, and much more wear out easily, and others become diminished due to the damage caused by WVCs. In the long run, implementing these proposed solutions leads to low costs as fewer properties are damaged, lost, and the frequency of animal carcasses minimized.
- Human prevention: The innovations proposed prevent lives of all the different stakeholders. Usually attained by minimizing death that frequently occurs due to collisions in hotspot areas. Stakeholders are such as insurance companies, highway agencies, citizens on social media, and any other persons that use road transport.
- Animal prevention: Preventing animals from road damage and death is vital. The proposed solutions ensure drivers are aware of the existence of animals on the road, which later limits collisions and minimizes the mortality rates of these animals.
- Environmental damage: The proposed solutions will minimize pollution caused by WVCs. The car collision parts and fuel smoke from cars pollute the environment. When minimizing pollution, climate change, and global warming, reduction in the extinction of animal species is limited.



Figure 8. Summary to justify the applicability of the proposed solutions.

The proposed systems in Section 7.2 contain machine learning functionalities. The functionalities aid in minimizing WVCs. In determining the practicability and feasibility of the proposed solutions, we propose and discuss a continuous general product development lifecycle in Figure 9. The workflow proposes steps to follow to determine the feasibility of machine learning systems, i.e., whether the project defined is feasible or not. To conduct a feasibility analysis for any machine learning project, product managers should follow the proposed workflow from product discovery, feasibility, product execution to product delivery. This workflow enables project managers to guide team members during the continuous process of product discovery to product delivery. In Figure 9, we describe the

product development lifecycle steps to follow when developing machine learning models and products. It has three continuous phases, product discovery, product execution, and product delivery, which are vital [147].

- Product Discovery: The discovery phase helps product managers to assess good ideas to consider for the final project. The output of the product discovery is the validated product backlog. The validated product backlog states four risks that every product needs to consider and address. Risks are tackled upfront rather than at the end. In modern teams, the risks are tackled before deciding on building anything. These risks include: (1) value risk, whether customers will buy the product or users will choose to use the product, (2) usability risk, whether users can figure out how to use the product, (3) feasibility risk, whether our engineers can build what we need with the time, skills, and technology we have, and (4) business viability risk, whether the solution also works for the various aspects of our business.
- Feasibility: The key objective is to evaluate all key factors relevant to a defined project after tackling the significant four risks. The components may be technical and non-technical, but all of these aid in predicting the likelihood of project success. When the non-technical factors are acceptable in the feasibility analysis, technical aspects are further processed and are modeled to quickly obtain a baseline model performance for the task defined while showing measurable progress on the selected baseline. The baseline model developed may rapidly use hyperparameter tuning to obtain different model architectures using an iterative process. If engineers work on a novel solution, the feasibility is more complex than upgrading and updating an existing system. Finally, the results from the feasibility analysis inform the decision on whether to proceed to the next step in the product lifecycle, i.e., product execution.
- Product Execution: Whenever the developed feasibility prototypes show that the performance would likely be acceptable, product managers engage the development team to execute the idea according to client needs. Products are defined and designed collaboratively rather than sequentially. In executing the project, the project team may follow the steps: (1) Define Design Requirements: After the feasibility analysis, project managers define the requirements of the system. A requirement is a need, functionality, or characteristic of a system [148]. A functional specification document describes the requirements to be implemented by the software solution. The document captures what the software needs to perform to support a user. The design requirements help product development teams prioritize work and decide what to build next. Following are two main types of requirements: (a) Functional requirements are those that must be met to deliver the project and are described from the customer's point of view, e.g., how will the customer experience it and/or benefit from it? Specific functional requirements may define each application in the machine learning domain. Some machine learning applications are credit card detection, whose functional requirement is a count of false positives and classification. The functional requirement for classification is a threshold on a count of low-confidence predictions that require human review and approval. (b) Non-functional/technical requirements define the system quality and determine the implementation of the system. Non-functional/technical requirements are system qualities such as security, accuracy, performance, security, data privacy, costs, availability, maintainability, operability, and scalability. (2) Model and Product Development: After defining design requirements, these, together with the selected aggregated data, act as input to the model and product development phase to build a baseline model or product. Later, evaluations of model output determine if the performance is acceptable. If yes, the model and product are deployed and sent to production. If no, a loop is activated to repeat the process from product discovery to product delivery
- Product Delivery: The process is continuous and enables delivery of executed products to the market, meeting the necessary performance, reliability, and fault. This phase

focuses on implementing a solution that solves an underlying problem. They typically focus on attaining the expected results.

- Production and Deployment: When the experiments are satisfactory, the model is deployed and set to production for clients to use.
- Support and Maintenance: This is the last step after model deployment and production. It can be challenging when conducting support and maintenance of machine learning products over time because machine learning is tightly coupled and integrated, meaning it depends on each of the components in the framework. Changes in one component, feature space, hyperparameters, and learning rate affect the model's performance. We need maintenance to track these issues and rectify them before they go astray.

Even though we find the proposed solutions novel, feasible, and also practical, they still need more case-by-case justification of the feasibility of each proposed solution. We, therefore, recommend future studies to verify the proposed systems' practicability and feasibility simply because feasibility analysis of a system is different for each project and location. It is essential to verify feasibility across different continents, countries, domains, weather conditions, types of animals, and road types. While verifying, the project teams may develop system prototypes and/or implement product development lifecycles for software and embedded systems.



Figure 9. The product development lifecycle for determining feasibility analysis of machine learning projects (adopted from Cagan [147]).

In summary, Section 7.1 discusses issues and challenges arising from the review, Section 7.2 proposes future research directions and a way forward, and Section 8 explains the applicability of the proposed solutions.

9. Conclusions

The emphasis on avoiding WVCs is a big concern in conservation. In mitigating WVCs, traditional methods such as underpasses, overpasses, plus fencing have been implemented and have contributed significantly to the decline of WVC mortality rates. Even when these traditional methods have recorded positive results in mitigating WVCs, they still have extreme limitations. These limitations have prompted the rise of ADSs such as area cover, mobile mapping, break-the-beam, and buried cable systems. These systems can even send an alert in case of intrusion in the detection area to caution the driver. However, these ADSs have also registered some limitations such as (1) failure to detect small animals, (2) false detection of animal species, and (3) ignorance of drivers on-road usage. A few works have introduced and implemented machine learning algorithms and datasets to help and overcome these limitations.

This paper aimed at generating rules that analyze how different factors contribute to WVCs and how the rules integrate machine learning to prevent WVCs. Previous solutions such as ADSs have contributed significantly to conservation research. In this review, we have criticized research on factors that influence WVCs, machine learning algorithms, ADSs, and datasets currently implemented to prevent WVCs. We have highlighted the issues and challenges arising from the review and proposed future solutions. We also noted that most of the current systems implemented to mitigate WVCs do not detect small animals. We suggest developing solutions in Section 7.2 to improve in detecting animals with a core goal of mitigating WVCs and further justify and determine the feasibility of the projects using the proposed continuous product development lifecycle.

Finally, we agree that animal detection and animal warning systems have contributed a positive impact on reducing WVCs. However, more research is needed to address the wildlife conditions in Africa. The automated systems should be developed and tested in Africa since Africa has different animal species and weather patterns than the Northern Hemisphere.

Author Contributions: Conceptualization, I.N. and M.A.; methodology, I.N., M.A. and P.O.; writing—original draft preparation, I.N.; writing—review and editing, M.A. and P.O.; supervision, M.A. and P.O. All authors have read and agreed to the published version of the manuscript.

Funding: This study was carried out with financial support from Rhodes University.

Institutional Review Board Statement: Ethical review and approval were waived for this study by Animal Research Ethics Committee of Rhodes University, South Africa, because the study did not involve the use of live animals and therefore did not require ethical clearance.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: We thank Gabi Teren and Wendy Collinson, who works with Endangered Wildlife Trust in South Africa, for the expert advice and valuable discussions. We would like to thank the anonymous reviewers for their comments that improved the paper.

Conflicts of Interest: The authors declare no conflict of interest.

References

- White, T. Defenders of Wildlife. Watch Out for Wildlife Facts. 2020. Available online: https://defenders.org/sites/default/files/ publications/collision_facts_and_figures.pdf (accessed on 5 June 2020).
- Tony, C.; Brian L.C.; Adam, F.; Marcel, H.; Bruce F.L.; Bethanie, W.; Chuck, W. Wildlife-Vehicle Collision Reduction Study: Report to Congress; Technical Report; U.S Department of Transportation: Washington, DC, USA, 2008.
- 3. Fatality Facts. Collisions with Fixed Objects and Animals. 2019. Available online: https://www.iihs.org/topics/fatality-statistics/detail/collisions-with-fixed-objects-and-animals (accessed on 3 November 2021).

- 4. Heigl, F.; Horvath, K.; Laaha, G.E.A. Amphibian and reptile road-kills on tertiary roads in relation to landscape structure: Using a citizen science approach with open-access land cover data. *BMC Ecol.* **2017**, *17*, 24. [CrossRef] [PubMed]
- 5. Almeida, A.; Azkune, G. Predicting human behaviour with recurrent neural networks. Appl. Sci. 2018, 8, 305. [CrossRef]
- Collision, W. Roadkill Study Highlights Wildlife Road Deaths. 2012. Available online: https://www.bridgestone.co.za/newsarticle/698/roadkill-study-highlights-wildlife-road-deaths (accessed on 23 April 2020).
- Kioko, J.; Kiffner, C.; Jenkins, N.; Collinson, W.J. Wildlife roadkill patterns on a major highway in northern Tanzania. *African Zool.* 2015, 50, 17–22. [CrossRef]
- 8. John, K.; Christian, K.; Payton, P.; Claire, P.A.; Wendy, C.; Samue, I.K. Driver Knowledge and Attitudes on Animal Vehicle Collisions in Northern Tanzania. *Trop. Conserv. Sci.* 2015, *8*, 352–366.
- Wheels24. Roadkill: Why So Many Animals Die on SA's Roads. 2015. Available online: https://www.wheels24.co.za/News/ Roadkill-in-SA-Distraction-not-speed-to-blame-20150422 (accessed on 9 April 2020).
- 10. Jakkula, V. Tutorial on Support Vector Machine (SVM); School of EECS, Washington State University: Pullman, WA, USA, 2006.
- 11. Lewis, D.D. Naive (Bayes) at forty: The independence assumption in information retrieval. In *European Conference on Machine Learning*; Springer: Berlin/Heidelberg, Germany, 1998; pp. 4–15.
- 12. Hassoun, M.H. Fundamentals of Artificial Neural Networks; MIT Press: London, UK, 1995.
- 13. Graves, A. Generating sequences with recurrent neural networks. *arXiv* 2013, arXiv:1308.0850.
- 14. Vedaldi, A.; Lenc, K. Matconvnet: Convolutional neural networks for matlab. In Proceedings of the 23rd ACM international conference on Multimedia, Brisbane, Australia, 26–30 October 2015; pp. 689–692.
- 15. Gers, F.A.; Schmidhuber, J.; Cummins, F. Learning to forget: Continual prediction with LSTM. *Neural Comput.* **2000**, *12*, 2451–2471. [CrossRef]
- 16. Gray, M. Advances in Wildlife Crossing Technologies. 2009. Available online: https://www.fhwa.dot.gov/publications/publicroads/09septoct/03.cfm (accessed on 1 July 2020).
- 17. Huijser, M.P.; McGowen, P.T.; Camel, W. Animal Vehicle Crash Mitigation Using Advanced Technology Phase I: Review, Design, and Implementation; Technical Report; Montana State University: Bozeman, MT, USA, 2006.
- Huijser, M.P.; Holland, T.D.; Blank, M.; Greenwood, M.C.; McGowen, P.T.; Hubbard, B.; Wang, S. *The Comparison of Animal Detection Systems in a Test-Bed: A Quantitative Comparison of System Reliability and Experiences with Operation and Maintenance;* Technical Report; Federal Highway Administration and Montana Department of Transportation: Helena, MT, USA, 2009.
- 19. Antônio, W.H.S.; Silva, M.D.; Miani, R.S.; Souza, J.R. A Proposal of an Animal Detection System Using Machine Learning. *Appl. Artif. Intell.* **2019**, *33*, 1093–1106. [CrossRef]
- 20. Huijser, M.P.; Hayden, L. *Evaluation of the Reliability of an Animal Detection System in a Test-Bed*; Technical Report; Western Transportation Institute: Berkeley, CA, USA, 2010.
- Smith, D.; Grace, M.; Miller, A.; Noss, M.; Noss, R. Assessing the Effectiveness and Reliability of the Roadside Animal Detection System on US Highway 41 Near the Turner River in Collier County; Technical Report, Contract No. BDV37, TWO #2; Florida Department of Transportation: Bartow, FL, USA, 2016.
- 22. Shapoval, V.; Lev, J.; Bartoška, J.; Kumhála, F. Application of Doppler Radar for Wildlife Detection in Vegetation. *Sci. Agric. Bohemica* **2018**, *49*, 136–141. [CrossRef]
- 23. Huijser, M.P.; Fairbank, E.R.; Abra, F.D. *The Reliability and Effectiveness of a Radar-Based Animal Detection System*; Technical Report; U.S. Department of Transportation: Washington, DC, USA, 2017.
- 24. Desholm, M. Thermal Animal Detection System (TADS). Development of a Method for Estimating Collision Frequency of Migrating Birds at Offshore Wind Turbines; Technical Report; National Environmental Research Institute: Copenhagen, Denmark, 2003.
- 25. Gordon, K.M.; McKinstry, M.C.; Anderson, S.H. Motorist response to a deer-sensing warning system. *Wildl. Soc. Bull.* 2004, 32, 565–573. [CrossRef]
- 26. Vikhram, B.; Revathi, B.; Shanmugapriya, R.; Sowmiya, S.; Pragadeeswaran, G. Animal detection system in farm areas. *Int. J. Adv. Res. Comput. Commun. Eng.* 2017, *6*, 587–591.
- 27. Mukherjee, A.; Sullivan, A.; Sinha, A.; Liu, X.; Brake, D. Roadway Monitoring and Driver Warning Systems for Wildlife-Vehicle Collision Avoidance; Technical Report; AUG Signals Ltd.: Toronto, ON, Canada, 2013.
- 28. Huijser, M.P.; Haas, C.; Crooks, K.R. *The Reliability and Effectiveness of an Electromagnetic Animal Detection and Driver Warning System*; Technical Report; Colorado Department of Transportation Research Branch: Bozeman, MT, USA, 2012.
- Druta, C.; Alden, A.S. Evaluation of a Buried Cable Roadside Animal Detection System; Technical Report; Virginia Center for Transportation Innovation and Research: Charlottesville, VA, USA, 2015.
- 30. Druta, C.; Alden, A.S. Preventing animal-vehicle crashes using a smart detection technology and warning system. *Transp. Res. Rec.* **2020**, 2674, 680–689. [CrossRef]
- 31. Sharma, S.; Shah, D. Real-time automatic obstacle detection and alert system for driver assistance on Indian roads. *Int. J. Veh. Autonom. Syst.* 2017, *13*, 189–202. [CrossRef]
- Rosenband, D.L. Inside Waymo's self-driving car: My favorite transistors. In Proceedings of the 2017 Symposium on VLSI Circuits, Kyoto, Japan, 5–8 June 2017; pp. C20–C22.
- 33. Sillero, N.; Hélder, R.; Marc, F.; Cristiano, S.; Gil, L. A road mobile mapping device for supervised classification of amphibians on roads. *Eur. J. Wildl. Res.* **2018**, *64*, 77. [CrossRef]

- Sousa Guedes, D.; Ribeiro, H.; Sillero, N. An Improved Mobile Mapping System to Detect Road-Killed Amphibians and Small Birds. *ISPRS Int. J. Geo-Inf.* 2019, 8, 565. [CrossRef]
- Lopes, G.; Ribeiro, A.F.; Sillero, N.; Gonçalves-Seco, L.; Silva, C.; Franch, M.; Trigueiros, P. High Resolution Trichromatic Road Surface Scanning with a Line Scan Camera and Light Emitting Diode Lighting for Road-Kill Detection. Sensors 2016, 16, 558. [CrossRef]
- Said, M.M.; Mohd, M.; Faye, I.; Husain, N.A.; Kamaruddin, T.T.; Dol, S. Review of Current Animal-Vehicle Collision (AVC) Studies. J. Soc. Automot. Eng. Malays. 2021, 5, 64–71.
- Pagany, R. Wildlife-vehicle collisions—Influencing factors, data collection and research methods. *Biol. Conserv.* 2020, 251, 108758.
 [CrossRef]
- Tibor, T.; Patrik, K.; Richard, O.; Miroslav, B.; Peter, S. Animal Recognition System Based on Convolutional Neural Network. *Adv. Electr. Electron. Eng.* 2017, 15, 517–525.
- Yue, S.; Bonebrake, T.C.; Gibson, L. Informing snake roadkill mitigation strategies in Taiwan using citizen science. *J. Wildl. Manag.* 2019, *83*, 80–88. [CrossRef]
- 40. Banupriyai, N.; Saranya, S.; Swaminathan, R.; Harikumar, S.; Palanisamy, S. Animal Detection Using Deep Learning Algorithm. *J. Crit. Rev.* **2020**, *7*, 434–439.
- Yi, J.Y.; Khot, R.A. ROOD: Unpacking the Design and the Making of a RoadKill Alert System. In Proceedings of the Fourteenth International Conference on Tangible, Embedded, and Embodied Interaction, Sydney, NSW, Australia, 9–12 February 2020; pp. 715–728.
- 42. Murphy, C.M. Writing an effective review article. J. Med. Toxicol. 2012, 8, 89–90. [CrossRef]
- 43. Khan, K.S.; Kunz, R.; Kleijnen, J.; Antes, G. Five steps to conducting a systematic review. J. R. Soc. Med. 2003, 96, 118–121. [CrossRef]
- 44. Ferrari, R. Writing narrative style literature reviews. *Med. Writ.* 2015, 24, 230–235. [CrossRef]
- 45. Perevochtchikova, M.; Flores, J.Á.H.; Marín, W.; Flores, A.L.; Bueno, A.R.; Negrete, I.A.R. Systematic review of integrated studies on functional and thematic ecosystem services in Latin America, 1992–2017. *Ecosyst. Serv.* **2019**, *36*, 100900. [CrossRef]
- Yang, K.; Meho, L.I. Citation analysis: A comparison of Google Scholar, Scopus, and Web of Science. Proc. Am. Soc. Inf. Sci. Technol. 2006, 43, 1–15. [CrossRef]
- 47. Ragab, O. Types of Roads & Road Compnents. Res. Gate 2016. [CrossRef]
- 48. Manan, M.M.A.; Várhelyi, A.; Çelik, A.K.; Hashim, H.H. Road characteristics and environment factors associated with motorcycle fatal crashes in Malaysia. *IATSS Res.* **2018**, *42*, 207–220. [CrossRef]
- Ken, S.; Richard, R.; Katherine, H. Gravel Roads. Construction & Maintenance Guide; Technical Report; U.S Department of Transportation, Federal Highway Administration: Tallahassee, FL, USA, 2015.
- Masino, J.; Thumm, J.; Levasseur, G.; Frey, M.; Gauterin, F.; Mikut, R.; Reischl, M. Characterization of road condition with data mining based on measured kinematic vehicle parameters. J. Adv. Transp. 2018, 2018, 8647607. [CrossRef]
- 51. Shepard, D.B.; Kuhns, A.R.; Dreslik, M.J.; Phillips, C.A. Roads as barriers to animal movement in fragmented landscapes. *Anim. Conserv.* **2008**, *11*, 288–296. [CrossRef]
- 52. van der Ree, R.; Gagnon, J.; Smith, D. Fencing: A Valuable Tool for Reducing Wildlife-Vehicle Collisions and Funnelling Fauna to Crossing Structures. In *Handbook of Road Ecology*, 1st ed.; Wiley Online Library: New York, NY, USA, 2015; pp. 159–171.
- 53. Juan, M.; Francisco, S.; Alberto, D. Can we mitigate animal-vehicle accidents using predictive models? *J. Appl. Ecol.* 2004, 41, 701–710.
- Carvalho, N.; Bordignon, M.; Shapiro1, J. Fast and furious: A look at the death of animals on the highway MS-080, Southwestern Brazil. *Iheringia Série Zool. Porto Alegre* 2014, 104, 43–49. [CrossRef]
- Collinson, W.; Davies-Mostert, H.; Davies-Mostert, W. Effects of culverts and roadside fencing on the rate of roadkill of small terrestrial vertebrates in northern Limpopo, South Africa. *Conserv. Evid.* 2017, 14, 39–43.
- Carvalho, C.F.; Custódio, A.E.I.; Marçal Júnior, O. Influence of climate variables on roadkill rates of wild vertebrates in the cerrado biome, Brazil. *Biosci. J.* 2017, 33, 1632–1641. [CrossRef]
- 57. Seiler, A.; Folkeson, L. *Habitat Fragmentation due to Transportation Infrastructure: Cost 341 National State-of-the-Art Report Sweden;* Technical Report; COST 341: Habitat Fragmentation Due to Transportation Infrastructure: Luxembourg, 2006.
- 58. Rytwinski, T.; Soanes, K.; Jaeger, J.A.G.; Fahrig, L.; Findlay, C.S.; Houlahan, J.; van der Ree, R.; van der Grift, E.A. How Effective Is Road Mitigation at Reducing Road-Kill? A Meta-Analysis. *PLoS ONE* **2016**, *11*, e0166941. [CrossRef] [PubMed]
- 59. Casado-Sanz, N.; Guirao, B.; Attard, M. Analysis of the Risk Factors Affecting the Severity of Traffic Accidents on Spanish Crosstown Roads: The Driver's Perspective. *Sustainability* **2020**, *12*, 2237. [CrossRef]
- 60. World Health Organization Europe. Fact Sheets on Sustainable Development Goals: Health Targets. 2017. Available online: https://www.euro.who.int/__data/assets/pdf_file/0003/351444/3.6-Fact-sheet-SDG-Road-safety-FINAL-10-10-2017.pdf (accessed on 28 June 2020).
- 61. Premti, A. Road Safety-Considerations in Support of the 2030 Agenda for Sustainable Development; Technical Report; United Nations: New York, NY, USA, 2018.
- 62. WHO. Road Safety: Basic Facts; Technical Report; World Health Organisation: Geneva, Switzerland, 2020.
- 63. Collier, P.; Conway, G.; Venables, T. Climate change and Africa. Oxf. Rev. Econ. Policy 2008, 24, 337–353. [CrossRef]
- 64. Short, F.T.; Neckles, H.A. The effects of global climate change on seagrasses. Aquat. Bot. 1999, 63, 169–196. [CrossRef]

- 65. Benson, N. Climate Change, Effects. In *Encyclopedia of Global Warming and Climate Change*; SAGE Publications, Inc.: New York, NY, USA, 2008; pp. 210–215.
- 66. Trenberth, K.E.; Miller, K.; Mearns, L.; Rhodes, S. *Effects of Changing Climate on Weather and Human Activities*; Technical Report; University Science Books: Sausalito, CA, USA, 2000.
- Change, N.G.C. The Effects of Climate Change. 2020. Available online: http://www.bexhillacademy.org/media/documents/ Physical%20Geog%20Starting%20Points%20Pages%2016%20-%2021.pdf (accessed on 29 June 2020).
- 68. Brennan, E. *Reducing the Impact of Global Warming on Wildlife: The Science, Management, and Policy Challenges Ahead;* Technical Report; Defenders of Wildlife: Washington, DC, USA, 2008.
- 69. Lister, N.M.; Brocki, M.; Ament, R. Integrated adaptive design for wildlife movement under climate change. *Front. Ecol. Environ.* **2015**, *13*, 493–502. [CrossRef]
- 70. Nora, B.E.A. *Wildlife in a Changing Climate;* Technical Report; Food and Agriculture Organisation of the United Nations Rome: Rome, Italy, 2012.
- 71. Changwan, S.; James, T.; Taeyoung, C.; Hyuksoo, K.; Chong-Hwa, P. Disentangling roadkill: The influence of landscape and season on cumulative vertebrate mortality in South Korea. *Landsc. Ecol. Eng.* **2013**, *11*, 87–99.
- 72. Bolhuis, J.J.; Giraldeau, L.A. The study of animal behavior. In *The Behavior of Animals: Mechanisms, Functions, and Evolutions;* Wiley: New York, NY, USA, 2005; pp. 1–10.
- 73. Ajzen, I. Understanding Attitudes and Predicting Social Behavior/Icek Ajzen, Martin Fishbein; Prentice-Hall: Hoboken, NJ, USA, 1980.
- 74. Fishbein, M.; Ajzen, I. Belief, Attitude, Intention, and Behavior. 1975. Available online: https://people.umass.edu/aizen/f&a197 5.html (accessed on 20 May 2020).
- 75. Ajzen, I. From Intentions to Action: A Theory of Planned Behavior; Springer: New York, NY, USA, 1975.
- McKenna, C.; Morrison, A. Scottish Government Social Research Group Social Science Methods Series. Guide 3: Natural Experiments. 2012. Available online: https://www2.gov.scot/resource/doc/175356/0091396.pdf (accessed on 30 August 2020).
- Nguyen, T.T.H.; Anderson, D.; Dunne, M.; Nguyen, H.T. Development and validation of a questionnaire to measure health professionals' attitudes toward identification of female victims of domestic violence. *Health* 2015, 7, 596–605. [CrossRef]
- Sheppard, B.H.; Hartwick, J.; Warshaw, P.R. The Theory of Reasoned Action: A Meta-Analysis of Past Research with Recommendations for Modifications and Future Research. J. Consum. Res. 1988, 15, 325–343. [CrossRef]
- 79. St John, F.; Edwards-Jones, G.; Jones, J. Conservation and human behaviour: Lessons from social psychology. *Wildl. Res.* **2010**, 37, 658–667. [CrossRef]
- 80. Labaw, P.J. Advanced Questionnaire Design; Abt Books: Cambridge, MA, USA, 1981.
- 81. Gendall, P. A Framework for Questionnaire Design: Labaw Revisited; Marketing Bulletin-Department of Marketing Massey University: Palmerston North, New Zealand, 1998; Volume 9, pp. 28–39.
- Gaston, G.; Gerjo, K. The Theory of Planned Behavior: A Review of its Applications to Health-Related Behaviors. Am. J. Health Promot. 1996, 11, 87–98.
- Academy, K. Elements of Behavior. 2020. Available online: https://www.khanacademy.org/science/biology/behavioralbiology/animal-behavior/a/intro-to-animal-behavior (accessed on 26 June 2020).
- 84. Nielsen, B.L.; de Jong, I.C.; De Vries, T.J. The use of feeding behaviour in the assessment of animal welfare. In *Nutrition and the Welfare of Farm Animals*; Springer: New York, NY, USA, 2016; pp. 59–84.
- 85. Sudesh Rathod. Habitant Selection. 2020. Available online: https://www.slideshare.net/sudeshrathod/habitat-selection (accessed on 27 June 2020).
- 86. Barki, A. Mating behaviour. In Reproductive Biology of Crustaceans; Science Publishers: New York, NY, USA, 2008; pp. 223–265.
- 87. Sumpter, D.J. The principles of collective animal behaviour. *Philos. Trans. R. Soc. B Biol. Sci.* **2006**, *361*, 5–22. [CrossRef]
- 88. Tierney, A. The evolution of learned and innate behavior: Contributions from genetics and neurobiology to a theory of behavioral evolution. *Anim. Learn. Behav.* **1986**, *14*, 339–348. [CrossRef]
- John, M. Elements of Behavior. 2016. Available online: https://projects.ncsu.edu/cals/course/ent425/library/tutorials/ behavior/elements_of_behavior.html (accessed on 26 June 2020).
- Tenney, S. Animal Behaviour. 2014. Available online: https://www.nature.com/scitable/knowledge/animal-behavior-13228230/ (accessed on 15 August 2020).
- 91. SAPeople. Who Said Animals Don't Roam Our Roads? 2016. Available online: https://www.sapeople.com/2016/06/09 /elephants-in-road-hoedspruit-wild-animals-south-africa/ (accessed on 27 June 2020).
- Hens, L.; Boon, E.K. Causes of Biodiversity Loss: A Human Ecological Analysis; Human Ecology Department, Vrije Universiteit Brussel: Brussels, Belgium, 2005.
- Shepard, E.L.; Wilson, R.P.; Quintana, F.; Laich, A.G.; Liebsch, N.; Albareda, D.A.; Halsey, L.G.; Gleiss, A.; Morgan, D.T.; Myers, A.E.; et al. Identification of animal movement patterns using tri-axial accelerometry. *Endangered Spec. Res.* 2008, 10, 47–60. [CrossRef]
- Abrahms, B.L. The Ecology and Conservation of Animal Movement in Changing Land-and Seascapes; University of California: Berkeley, CA, USA, 2016.
- 95. Urbano, F.; Cagnacci, F.; Calenge, C.; Dettki, H.; Cameron, A.; Neteler, M. Wildlife tracking data management: A new vision. *Philos. Trans. R. Soc. B Biol. Sci.* 2010, 365, 2177–2185. [CrossRef]

- 96. Thomas, B.; Holland, J.D.; Minot, E.O. Wildlife tracking technology options and cost considerations. *Wildl. Res.* **2011**, *38*, 653–663. [CrossRef]
- Handcock, R.N.; Swain, D.L.; Bishop-Hurley, G.J.; Patison, K.P.; Wark, T.; Valencia, P.; Corke, P.; O'Neill, C.J. Monitoring animal behaviour and environmental interactions using wireless sensor networks, GPS collars and satellite remote sensing. *Sensors* 2009, 9, 3586–3603. [CrossRef]
- 98. Wildlife ACT. GPS and VHF Tracking Collars used for Wildlife Monitoring. *Wildlife ACT News*, 17 April 2014. Available online: https://wildlifeact.com/blog/gps-and-vhf-tracking-collars-used-for-wildlife-monitoring/ (accessed on 24 June 2020).
- 99. Emslie, K. 5 Reasons why Wildlife ACT Collar and Monitor Wild Dogs. *Wildlife ACT News*, 4 October 2012. Available online: https://wildlifeact.com/blog/reasons-why-wildlife-act-collar-monitor-wild-dogs/ (accessed on 24 June 2020).
- 100. Spink, A.; Cresswell, B.; Kölzsch, A.; Van Langevelde, F.; Neefjes, M.; Noldus, L.; Van Oeveren, H.; Prins, H.; Van Der Wal, T.; De Weerd, N.; et al. Animal behaviour analysis with GPS and 3D accelerometers. In Proceedings of the 6th European Conference on Precision Livestock Farming, Leuven, Belgium, 10–12 September 2013; pp. 229–239.
- 101. Arac, A.; Zhao, P.; Dobkin, B.H.; Carmichael, S.T.; Golshani, P. DeepBehavior: A Deep Learning Toolbox for Automated Analysis of Animal and Human Behavior Imaging Data. *Front. Syst. Neurosci.* **2019**, *13*, 20. [CrossRef]
- Perner, P. Motion tracking of animals for behavior analysis. In *International Workshop on Visual Form*; Springer: Berlin/Heidelberg, Germany, 2001; pp. 779–786.
- 103. de Weerd, N.; van Langevelde, F.; van Oeveren, H.; Nolet, B.A.; Kölzsch, A.; Prins, H.H.T.; de Boer, W.F. Deriving Animal Behaviour from High-Frequency GPS: Tracking Cows in Open and Forested Habitat. *PLoS ONE* **2015**, 10, e0129030. [CrossRef]
- 104. Valletta, J.J.; Torney, C.; Kings, M.; Thornton, A.; Madden, J. Applications of machine learning in animal behaviour studies. *Anim. Behav.* 2017, 124, 203–220. [CrossRef]
- 105. Kaur, M.; Randhawa, R. Animal Detection: Techniques, Challenges and Future Scope. *Int. J. Innov. Technol. Explor. Eng.* **2019**, *9*, 4706–4710.
- 106. Smietan, I. *Perimeter Security Sensor Technologies Handbook*; Technical Report; U.S Department of Justice, Office of Justice Programs: Rockville, MD, USA, 1997.
- 107. Shwetha, B.; Nitesh, M.; Abhishek, C. Passive Infrared (PIR) Sensor Based Security Control System using Microcontroller using 89C51. In Proceedings of the CCSO 2013, 2013; pp. 93–96. Available online: http://www.conference.bonfring.org/papers/ait_ ccso2013/ccso056.pdf (accessed on 20 October 2021).
- 108. Huijser, M.P.; McGowen, P.T. Overview of animal detection and animal warning systems in North America and Europe. In Proceedings of the 2003 International Conference on Ecology and Transportation, Raleigh, NC, USA, 24 August 2003; Irwin, C.L., Garrett, P., McDermott, K.P., Eds.; Center for Transportation and the Environment, North Carolina State University: Raleigh, NC, USA, 2003; pp. 368–382.
- 109. Huijser, M.P.; McGowan, P.; Hardy, A.; Kociolek, A.; Clevenger, A.; Smith, D.; Ament, R. Wildlife-Vehicle Collision Reduction Study: Report to Congress; Technical Report; Western Association of Fish and Wildlife Agencies: Washington, DC, USA, 2017.
- Goswami, M.; Prakash, V.P.; Goswami, D. Animal-Vehicle Collision Mitigation Using Deep Learning in Driver Assistance Systems. In Proceedings of the International Conference on Advances in Computing and Data Sciences, Ghaziabad, India, 12–13 April 2019; pp. 284–295.
- Petrović, D.; Mijailović, R.; Pešić, D. Traffic accidents with autonomous vehicles: Type of collisions, manoeuvres and errors of conventional vehicles' drivers. *Transp. Res. Procedia* 2020, 45, 161–168. [CrossRef]
- 112. Valerio, F.; Basile, M.; Balestrieri, R. The identification of wildlife-vehicle collision hotspots: Citizen science reveals spatial and temporal patterns. *Ecol. Process.* **2021**, *10*, 6. [CrossRef]
- 113. Rowden, P.; Steinhardt, D.; Sheehan, M. Road crashes involving animals in Australia. *Accid. Anal. Prev.* 2008, 40, 1865–1871. [CrossRef]
- 114. Lala, F.; Chiyo, P.I.; Kanga, E.; Omondi, P.; Ngene, S.; Severud, W.J.; Morris, A.W.; Bump, J. Wildlife roadkill in the Tsavo Ecosystem, Kenya: Identifying hotspots, potential drivers, and affected species. *Heliyon* **2021**, *7*, e06364. [CrossRef]
- 115. Drews, C. Road kills of animals by public traffic in Mikumi National Park, Tanzania, with notes on baboon mortality. *Afr. J. Ecol.* **1995**, *33*, 89–100. [CrossRef]
- 116. Périquet, S.; Roxburgh, L.; le Roux, A.; Collinson, W.J. Testing the Value of Citizen Science for Roadkill Studies: A Case Study from South Africa. *Front. Ecol. Evol.* **2018**, *6*, 15. [CrossRef]
- 117. Sachin, P. Pothole Image Data-Set. 2019. Available online: https://www.kaggle.com/sachinpatel21/pothole-image-dataset (accessed on 20 October 2021).
- 118. Jessica, L. Stanford Cars Dataset. 2018. Available online: https://www.kaggle.com/jessicali9530/stanford-cars-dataset (accessed on 20 October 2021).
- 119. David, M. Oregon Wildlife. 2019. Available online: https://www.kaggle.com/virtualdvid/oregon-wildlife (accessed on 20 October 2021).
- Fan, T.; Sadeghian, R.; Aram, S. Deer-vehicle collisions prevention using deep learning techniques. In Proceedings of the 2020 IEEE Cloud Summit, Cloud Summit 2020, Harrisburg, PA, USA, 21–22 October 2020; pp. 97–102.
- 121. Willi, M.; Pitman, R.T.; Cardoso, A.W.; Locke, C.; Swanson, A.; Boyer, A.; Veldthuis, M.; Fortson, L. Identifying animal species in camera trap images using deep learning and citizen science. *Methods Ecol. Evol.* **2019**, *10*, 80–91. [CrossRef]

- Norouzzadeh, M.S.; Nguyen, A.; Kosmala, M.; Swanson, A.; Palmer, M.S.; Packer, C.; Clune, J. Automatically identifying, counting, and describing wild animals in camera-trap images with deep learning. *Proc. Natl. Acad. Sci. USA* 2018, 115, E5716–E5725. [CrossRef] [PubMed]
- 123. Farhadi, A.; Redmon, J. Yolov3: An incremental improvement. arXiv 2018, arXiv:1804.02767.
- 124. Sun, P.; Kretzschmar, H.; Dotiwalla, X.; Chouard, A.; Patnaik, V.; Tsui, P.; Guo, J.; Zhou, Y.; Chai, Y.; Caine, B.; et al. Scalability in perception for autonomous driving: Waymo open dataset. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Seattle, WA, USA, 14–19 June 2020; pp. 2446–2454.
- 125. Backs, J.; Nychka, J.; Clair, C.S. Warning systems triggered by trains could reduce collisions with wildlife. *Ecol. Eng.* 2017, 106, 563–569. [CrossRef]
- 126. Geiger, A.; Lenz, P.; Stiller, C.; Urtasun, R. Vision meets robotics: The kitti dataset. *Int. J. Robot. Res.* 2013, 32, 1231–1237. [CrossRef]
- 127. Parikh, M.; Patel, M.; Bhatt, D. Animal detection using template matching algorithm. *Int. J. Res. Mod. Eng. Emerg. Technol.* 2013, 1, 26–32.
- 128. Zhang, Y.; Wang, X.; Qu, B. Three-frame difference algorithm research based on mathematical morphology. *Procedia Eng.* 2012, 29, 2705–2709. [CrossRef]
- 129. Wold, S.; Esbensen, K.; Geladi, P. Principal Component Analysis. Chimometr. Intell. Lab. Syst. 1987, 2, 37–52. [CrossRef]
- 130. Balakrishnama, S.; Ganapathiraju, A. Linear discriminant analysis—A brief tutorial. Inst. Signal Inf. Process. 1998, 18, 1-8.
- 131. Ojala, T.; Pietikäinen, M.; Harwood, D. A comparative study of texture measures with classification based on featured distributions. *Patt. Recognit.* **1996**, *29*, 51–59. [CrossRef]
- Simonyan, K.; Zisserman, A. Very deep convolutional networks for large-scale image recognition. In Proceedings of the 3rd International Conference on Learning Representations, ICLR 2015-Conference Track Proceedings, San Diego, CA, USA, 7–9 May 2015.
- He, K.; Zhang, X.; Ren, S.; Sun, J. Deep residual learning for image recognition. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, NV, USA, 27–30 June 2016; pp. 770–778.
- 134. Szegedy, C.; Vanhoucke, V.; Ioffe, S.; Shlens, J.; Wojna, Z. Rethinking the inception architecture for computer vision. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, NV, USA, 27–30 June 2016; pp. 2818–2826.
- 135. Chollet, F. Xception: Deep learning with depthwise separable convolutions. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, HI, USA, 21–26 July 2017; pp. 1251–1258.
- 136. Phillips, S.J.; Anderson, R.P.; Schapire, R.E. Maximum entropy modeling of species geographic distributions. *Ecol. Model.* 2006, 190, 231–259. [CrossRef]
- Yu, H.; Shen, Z.; Miao, C.; Leung, C.; Lesser, V.R.; Yang, Q. Building Ethics into Artificial Intelligence. In Proceedings of the Twenty-Seventh International Joint Conference on Artificial Intelligence, IJCAI-18, Stockholm, Sweden, 13–19 July 2018; pp. 5527–5533.
- 138. Wearn, O.R.; Freeman, R.; Jacoby, D.M. Responsible AI for conservation. Nat. Mach. Intell. 2019, 1, 72–73. [CrossRef]
- 139. Khandelwal, P. Movebank: Animal Tracking. Analyzing Migratory Patterns of Animals. 2017. Available online: https://www.kaggle.com/pulkit8595/movebank-animal-tracking (accessed on 24 June 2020).
- 140. Wang, G. Machine learning for inferring animal behavior from location and movement data. *Ecol. Inform.* **2019**, *49*, 69–76. [CrossRef]
- 141. Brown, S.; Davidovic, J.; Hasan, A. The algorithm audit: Scoring the algorithms that score us. *Big Data Soc.* 2021, *8*, 2053951720983865. [CrossRef]
- 142. Liu, X.; Xie, L.; Wang, Y.; Zou, J.; Xiong, J.; Ying, Z.; Vasilakos, A.V. Privacy and security issues in deep learning: A survey. *IEEE Access* 2020, *9*, 4566–4593. [CrossRef]
- 143. Raza, A.; Tran, K.P.; Koehl, L.; Li, S. Designing ecg monitoring healthcare system with federated transfer learning and explainable ai. *Knowl.-Based Syst.* 2022, 236, 107763. [CrossRef]
- 144. Manresa-Yee, C.; Ramis, S. Assessing Gender Bias in Predictive Algorithms using eXplainable AI. In Proceedings of the XXI International Conference on Human Computer Interaction, Málaga, Spain, 22–24 September 2021; pp. 1–8.
- 145. Oueida, S.; Hossain, S.Q.; Kotb, Y.; Ahmed, S.I. A Fair and Ethical Healthcare Artificial Intelligence System for Monitoring Driver Behavior and Preventing Road Accidents. In Proceedings of the Future Technologies Conference, Vancouver, BC, Canada, 28–29 October 2021; Springer: Cham, Switzerland, 2021; pp. 431–444.
- 146. Irene, N.; Marcellin, A.; Patrice, O. Integrating AI ethics in wildlife conservation AI systems in South Africa: A review, challenges, and future research agenda. *AI Soc.* **2021**, 1–13. [CrossRef]
- 147. Cagan, M. Inspired: How to Create Products Customers Love; SVPG Press: Sunnyvale, CA, USA, 2008.
- 148. *IEEE Std 610.12-1990;* IEEE Standard Glossary of Software Engineering Terminology. IEEE Computer Society Standards Coordinating Committee: Piscataway, NJ, USA, 1990.